Power Electronic Transformers for AC-AC and AC-DC Conversion with Reduced Number of Switches

A DISSERTATION SUBMITTED TO THE FACULTY OF THE GRADUATE SCHOOL OF THE UNIVERSITY OF MINNESOTA

 $\mathbf{B}\mathbf{Y}$

Gysler Fatima Castelino

IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF **Doctor of Philosophy**

Professor Ned Mohan

August, 2013

© Gysler Fatima Castelino 2013 ALL RIGHTS RESERVED

Acknowledgements

I would like to express my deepest gratitude to my advisor, Professor Mohan, firstly for giving me this opportunity to work in his research group and for guiding me through my graduate studies. It has been an honor to be Professor Mohan's student and teaching assistant. Without his valuable advice, excellent teaching and encouragement, this thesis would not have been possible.

I would like to thank my committee members, Professor Wollenberg, Professor Robbins and Professor Bobkov for their support and for being part of my PhD exam committee.

I would like to thank my colleagues at the lab who have knowingly or unknowingly helped me become a better engineer - Ranjan and Dr. Krushna Mohaparta for getting me started on this innovative project, Hari for teaching me to debug circuits, Apurva for teaching me to believe in myself, Kaushik for teaching me how to write technical papers, Saurabh for always being there to bounce ideas off and Dr. Chris Henze for his invaluable practical advice. I have learned so much from Dr. Dinkar Prasad, Dr. Amod Umrikar, Eric, Shanker, Rohit, Nathan, David, Shabri, Rashmi, John, Mudita, Tamil, Ruben, Siddharth, Viswesh and Santhosh. Thank you for providing a great lab in which to work, learn and grow together.

This work is supported by Office of Naval Research, Grant N00014-11-1-0897 and Department of Energy Award Number DE-EE0002980. Their financial support is gratefully acknowledged.

I am very blessed to be surrounded by family and friends who have wholeheartedly supported and encouraged all of my endeavors. I am grateful to Sanmitra, Shruti, Gayle, Smita, Brianna, Keith, Paresh, DK, Dani, Satyakant, Sr. Mark, Abhijit, Mukta, Puskhar and Harshada for always being there for me. Last but not the least, a very special thanks to my loving parents Rudolph and Lydia, my brother Kenneth, my sister Lou-ann, my godma Maria and my husband Benjamin and his family for their unconditional love.

Dedication

To my family near and far, my friends and my teachers.

Abstract

Power Electronic Transformers (PETs) operating at frequencies well above the grid frequency have advantages of reduced size and weight as compared to their grid frequency counterparts. They have been proposed for future distribution systems as well as for use in adjustable speed drives where space and weight are at a premium. Power Electronic Transformers are especially attractive to the Navy in power converters for propulsion, radar, lighting and other needs in submarines and ships where the size, cost and weight of power electronic converters needs to be low. Furthermore, these high power density converters have been proposed for harnessing wind energy, especially offshore wind resources.

In this research, a novel reduced-switch Power Electronic Transformer is proposed for three-phase power conversion (three-phase AC to AC as well as three-phase AC to DC). The goal of this project is to reduce the weight and size of the power converter by replacing the low-frequency transformers with high-frequency transformers (HFTs). The novelty of these proposed topologies is that they have only two controlled switches on the primary side of the high-frequency transformers. Additionally, these switches operate at 50% duty ratio, hence they are easy to control. Pulse Width Modulation (PWM) control is only necessary in the converters on the secondary side of the HFT. The modulation strategies proposed in this work achieve nearly Zero Current Switching (ZCS) for these two primary switches.

In the proposed three-phase AC to AC Power Electronic Transformer, a Matrix Converter (MC) is employed on the secondary side of the transformer. Matrix converter with nine four-quadrant switches is a 'more-silicon' and nearly capacitor-less solution for AC to AC conversion. This single-stage AC-AC converter without any electrolytic capacitors translates to higher reliability and efficiency. High-frequency transformers have finite leakage inductance, hence, any switching in the primary or secondary of the transformer requires commutation of the transformer inductive currents. This problem of leakage energy commutation and the required protection are studied in detail in this thesis. Other features of this converter are bi-directional power flow and power factor correction.

The second Power Electronic Transformer is proposed for three-phase AC to DC power conversion. This converter operates on the Dual Active Bridge (DAB) principle wherein the transformer leakage inductance is used for power transfer. Hence, this converter does not suffer from problems associated with leakage energy commutation and additional snubber circuits are not required for this converter. The proposed modulation provides the advantages of unity power factor on the AC side, galvanic isolation and bi-directional power flow capability.

Both these PETs have been analyzed and simulated. Laboratory prototypes have been built and tested to verify the advantages of the proposed PETs.

Contents

A	ckno	wledgements	i
D	edica	tion	iii
A	bstra	let	\mathbf{iv}
Li	st of	Tables	x
Li	st of	Figures	xii
1	Intr	roduction	1
	1.1	Example applications for PETs	2
		1.1.1 Power Electronic Transformers for harnessing wind energy	2
		1.1.2 Power Electronic Transformers in PHEVs	4
	1.2	Three-phase AC-AC Power Electronic Transformers	4
	1.3	Three-phase AC-DC Power Electronic Transformer	6
	1.4	Contributions of this thesis	7
	1.5	Organization of this thesis	8
2	AC	AC Power Electronic Transformer: Topology and Modulation	9
	2.1	Introduction	9
	2.2	PET topology and modulation strategy	11
	2.3	A PWM technique to eliminate the input clamp circuit	15
	2.4	Simulation results	19
	2.5	Experimental setup	19

		2.5.1	Transformer voltages	21
	2.6	Exper	imental results: Carrier based modulation	23
	2.7	Exper	imental results: Space vector modulation	25
		2.7.1	Power factor control	27
		2.7.2	Fourier analysis of input currents	28
		2.7.3	Soft-switching of primary converter	30
		2.7.4	Output current and voltage	30
		2.7.5	Extended power factor control	30
		2.7.6	Variable frequency generation	33
	2.8	Conclu	usions and future work	33
		2.8.1	Future work	35
3	AC	AC P	ower Electronic Transformer: Clamp Circuit Analysis	36
	3.1	Introd	luction	36
	3.2	Introd	luction of the secondary clamp circuit	37
	3.3	Clamp	p circuit analysis	38
		3.3.1	Active to zero vector transition $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	40
		3.3.2	Zero vector to active vector transition $\ldots \ldots \ldots \ldots \ldots \ldots$	43
	3.4	Simula	ation results	46
		3.4.1	Intervals t_{az1} and t_{za1}	46
		3.4.2	Three-phase AC-AC case for same frequency	48
		3.4.3	Three-phase variable frequency AC	49
	3.5	Exper	imental results	49
	3.6	Comp	arison of dead-time commutation and four-step commutation	50
		3.6.1	Simulation results	53
		3.6.2	Experimental results	54
	3.7	Conclu	usions	58
4	$\mathbf{Sin}_{\mathbf{i}}$	gle-Pha	ase AC-DC Power Electronic Transformer	60
	4.1	Topole	ogy and modulation	61
		4.1.1	Analysis of DC-DC converter	62
		4.1.2	Analysis of single-phase AC-DC converter	66
	4.2	Simula	ation results	69

	4.3	Experimental setup	72
	4.4	Experimental results: DC-DC converter	73
		4.4.1 Effects of non-idealities in the circuit $\ldots \ldots \ldots \ldots \ldots \ldots$	77
	4.5	Experimental results: Single-phase AC-DC converter	79
		4.5.1 Effects of non-idealities in the circuit	81
	4.6	Conclusions and future work	83
		4.6.1 Future work	87
5	Thr	ee-phase AC-DC Power Electronic Transformer	88
	5.1	Topology and modulation technique	89
	5.2	Analysis	94
	5.3	Simulation results	97
	5.4	Closed-loop control of three-phase AC-DC converter $\hdots \ldots \ldots \ldots$.	99
	5.5	Experimental results	102
	5.6	Conclusions and future work	109
		5.6.1 Future Work	109
6	Cor	nclusion	110
R	efere	nces	112
A	ppe	ndix A. Acronyms	122
A	ppe	ndix B. Experimental Setup	123
	B.1	Matrix converter board	123
		B.1.1 Clamp circuits	123
	B.2	Two-level converter	125
	B.3	Primary circuit	125
	B.4	Gate driver	126
	B.5	Sensing and control	126
A	ppe	ndix C. Transformer Design	127

Apper	ndix D. FPGA Code for AC-AC Power Electronic Transformer	130
D.1	Carrier Based Modulation	132
D.2	Space Vector Modulation	133
D.3	Commutation	134
Apper	ndix E. FPGA Code for AC-DC Power Electronic Transformer	136
E.1	Single-phase AC-DC Power Electronic Transformer	136
E.2	Three-phase AC-DC Power Electronic Transformer	137

List of Tables

2.1	Simulation parameters	18
2.2	Experimental setup: parameters	21
2.3	Experimental results: Carrier based modulation	25
2.4	Power factor control	28
2.5	Extended power factor control	33
3.1	Commutation time (active to zero, S_1 on)	42
3.2	Commutation time for zero to active vector $\ldots \ldots \ldots \ldots \ldots \ldots$	45
3.3	Simulation parameters	47
3.4	Simulation results	47
3.5	Simulation parameters: Three-phase system	48
3.6	Experimental setup: Parameters	50
3.7	Experimental results: Zero to active vector transition	51
3.8	Experimental results: Active to zero vector transition $\ldots \ldots \ldots \ldots$	52
3.9	Experimental parameters: Comparison of commutation techniques \ldots	57
3.10	Comparison of modulation and commutation $\ldots \ldots \ldots \ldots \ldots$	58
4.1	Simulation parameters	70
4.2	Simulation results	71
4.3	Experimental parameters	75
4.4	DC-DC converter: Experimental results	79
4.5	Single-phase AC-DC converter: Experimental results	83
5.1	Simulation and experimental parameters for three-phase AC-DC converter	97
5.2	Analytical and simulated values	99
5.3	Simulation parameters: Closed-loop control	100
5.4	Experimental and analytical results of P_{io}	105

5.5	Experimental and analytical results of RMS currents	105
C.1	AC-AC PET operating conditions	127
C.2	Transformer design parameters	129
C.3	Designed transformer	129
D.1	Sector Transformation	134

List of Figures

1.1	Classification of Power Electronic Transformers	2
1.2	Conventional wind turbine systems	3
1.3	Wind turbine systems with low-frequency transformer in the nacelle $\$.	3
1.4	Proposed Power Electronic Transformer for wind turbine	3
1.5	First Power Electronic Transformer [1]	4
1.6	Multi-stage Power Electronic Transformer	5
2.1	Proposed AC-AC PET topology	11
2.2	Switching pulses for S_1 and S_2	13
2.3	Primary clamp circuit analysis	13
2.4	CCW rotating vectors	17
2.5	CW rotating vectors	17
2.6	Switching pulses for new modulation technique	18
2.7	Simulation results I	19
2.8	Simulation results II	20
2.9	Simulation results III	20
2.10	Experimental setup	22
2.11	Experimental setup for PET for AC to AC power conversion $\ldots \ldots$	22
2.12	Experimental results: Transformer voltages	23
2.13	Experimental results for carrier based modulation I	24
2.14	Experimental results for carrier based modulation II $\ldots \ldots \ldots \ldots$	25
2.15	Experimental results for space vector modulation	26
2.16	Real and reactive power of the PET	27
2.17	Experimental results: Fourier spectrum	29
2.18	Experimental results: Transformer currents	31

2.19	Experimental results: SVM	32	
2.20	Extended power factor control	33	
2.21	Experimental results: Extended power-factor control $\ldots \ldots \ldots \ldots$	34	
2.22	Power factor variation for different values of β	34	
3.1	Topology with non-idealities	37	
3.2	Switching pulses	38	
3.3	Active vector to zero vector equivalent circuit	40	
3.4	Clamp circuit power loss : Active to zero vector	42	
3.5	Clamp circuit power loss: Active to zero vector $\ldots \ldots \ldots \ldots \ldots$	42	
3.6	Commutation time required: Active to zero vector $\ldots \ldots \ldots \ldots$	43	
3.7	Zero vector to active vector equivalent circuit $\ldots \ldots \ldots \ldots \ldots$	44	
3.8	Clamp circuit power loss : Zero to active vector $\ldots \ldots \ldots \ldots$	45	
3.9	Clamp circuit power loss : Zero to active vector $\ldots \ldots \ldots \ldots$	45	
3.10	Commutation time required: Zero to active vector $\ldots \ldots \ldots \ldots$	46	
3.11	Active to zero vector simulation results (t_{az1})	47	
3.12	Zero to active vector simulation results (t_{za1})	47	
3.13	Simulation results	48	
3.14	1kW, 0.8 pf, three-phase variable frequency 60Hz - 120Hz $\ .$	49	
3.15	Experimental setup for active to zero vector transition		
3.16	Clamp circuit analysis: Experimental results I	51	
3.17	Clamp circuit analysis: Experimental results II $\ldots \ldots \ldots \ldots \ldots$	52	
3.18	Experimental comparison of dead-time and four-step commutation I $$	54	
3.19	Experimental comparison of dead-time and four-step commutation II	55	
3.20	Experimental comparison of dead-time and four-step commutation III $$.	56	
3.21	Experimental comparison of dead-time and four-step commutation $\operatorname{IV}\;$.	58	
4.1	Single-phase AC-DC PET	61	
4.2	Equivalent circuit of single-phase AC-DC converter	63	
4.3	Switching waveforms for DC-DC converter	64	
4.4	DC-DC Converter: P_t Vs P_{io}	66	
4.5	DC-DC Converter: \overline{I}_{rpl} Vs δ	66	
4.6	AC-DC Converter: P_{avg} Vs δ	68	
4.7	AC-DC Converter: P_t Vs P_{avg}	69	

4.8	AC-DC Converter: \overline{I}_{rpl} Vs δ	69
4.9	Simulation result: DC-DC converter	70
4.10	Simulation result: Single-phase AC-DC	71
4.11	Single-phase AC-DC converter: Primary circuit	72
4.12	Experimental setup: schematic	74
4.13	Experimental setup:picture	74
4.14	DC-DC converter: Experimental results-I	75
4.15	DC-DC converter: Experimental results-II	76
4.16	DC-DC converter: Experimental results-III	76
4.17	DC-DC converter: Experimental results-IV	77
4.18	DC-DC converter: Experimental results-V	78
4.19	DC-DC converter: Experimental results-VI	78
4.20	DC-DC converter: Experimental results-VII	79
4.21	Single-phase AC-DC converter: Experimental results-I	80
4.22	Single-phase AC-DC converter: Experimental results-II	81
4.23	Single-phase AC-DC converter: Experimental results-III	82
4.24	Single-phase AC-DC converter: Experimental results-IV	82
4.25	Single-phase AC-DC converter: Experimental results-V	83
4.26	Single-phase AC-DC converter: Experimental results-VI	84
4.27	Single-phase AC-DC converter: Experimental results-VII	85
4.28	Single-phase AC-DC converter: Experimental results-VIII	85
4.29	Single-phase AC-DC converter: Experimental results-IX	86
5.1	Multi-stage Power Electronic Transformer for AC to DC conversion	88
5.2	Three-phase AC-DC converter (PET): Topology	90
5.3	Two-level converter space vectors	91
5.4	Switching waveforms	93
5.5	Three-phase AC-DC: P_{io} Vs δ	94
5.6	Three-phase AC-DC: \overline{I}_{rpl} Vs δ	96
5.7	Three-phase AC-DC: P_t Vs P_{io}	96
5.8	Three-phase AC-DC: Simulation I	98
5.9	Three-phase AC-DC: Simulation II	98
5.10	Three-phase AC-DC: Simulation III	98

5.11	Three-phase AC-DC: Simulation IV	98	
5.12	Three-phase AC-DC: Control I	99	
5.13	Three-phase AC-DC: Control II		
5.14	Three-phase AC-DC: Control III	101	
5.15	Three-phase AC-DC: Control IV	101	
5.16	Three-phase AC-DC: Control V	101	
5.17	Schematic of experimental setup	103	
5.18	Photograph of experimental setup	104	
5.19	Three-phase AC-DC: Experimental results I	104	
5.20	Three-phase AC-DC: Experimental results II	106	
5.21	Three-phase AC-DC: Experimental results III	107	
5.22	Three-phase AC-DC: Experimental results IV	107	
5.23	Three-phase AC-DC: Experimental results V $\ldots \ldots \ldots \ldots \ldots$	108	
B.1	Matrix converter board schematic	124	
B.2	Matrix converter board	124	
B.3	Matrix converter board configured as a two-level inverter $\hfill \ldots \ldots \ldots$	125	
B.4	Primary circuit: PET for AC-AC conversion	125	
B.5	FPGA control board	126	
C.1	Three-winding transformer	127	
C.2	Picture of the three-winding transformers	129	
D.1	FPGA algorithm: AC-AC PET	131	
D.2	Stationary reference frame	132	
D.3	SVM: Sector Determination	134	
D.4	Four-step commutation pulses	135	
E.1	FPGA implementation: Single-phase AC-DC PET	137	
E.2	FPGA algorithm: Three-phase AC-DC PET	137	

Chapter 1

Introduction

Power transformers are building-blocks of our power grid. They are indispensable because they provide voltage transformation that enables power transmission and galvanic isolation for protection. However, these power transformers suffer from some potential limitations: 1) heavy weight because of the size of magnetics 2) use of mineral oils that are not environmentally friendly and 3) sensitivity to harmonics [2]. The size of a transformer depends on its operating frequency, saturation flux density of the core material and the thermal considerations of the core and winding. Operating a transformer at frequencies above the grid frequency has been shown to have advantages of lower weight and higher power density. The use of magnetic materials such as FINEMET with high saturation flux density further reduces the size and weight of High-frequency Transformers (HFTs) [3].

With the help of power electronics, bulky line frequency (50Hz, 60Hz) transformers can be replaced with compact high-frequency transformers. This concept of Power Electronic Transformer (PET) was first introduced by McMurray in 1968 [1]. It has been explored by many researchers and is also called Solid-State Transformer (SST) [4] and Intelligent Universal Transformer. The key properties of PETs are high-frequency isolation along with voltage transformation, voltage regulation, bi-directional power flow capability and in some PETs, reactive power control can be achieved. The definition of PETs has extended to AC to DC conversion as well. The classification of Power Electronic Transformers (PET) is given in Fig. 1.1.



Figure 1.1: Classification of Power Electronic Transformers

PETs require a large number of power semiconductor devices. However, rapid advances in power electronic device technology have made available devices that have high voltage and current ratings with good switching characteristics as well as the ability to perform better at high temperatures [5]. This has played an important role in renewing interest in PETs.

In this chapter, an application of PETs for wind energy and for Plug-in Hybrid Vehicles is described followed by a summary of the state-of-the-art PETs for threephase AC-AC and AC-DC power conversion.

1.1 Example applications for PETs

Distributed generation in the form of renewable energy resources has increased manyfold in the past decade and it is predicted to increase even more in the years to come. Power electronics provide the enabling technology to connect these variable distributed resources to the power grid that operates at a constant frequency (50Hz, 60Hz). The variability of renewable resources make storage an important element of the modern power grid. Storage can be achieved using batteries, fuel cells, flywheels and recently even plug-in hybrid electric vehicles have been suggested for energy storage. Here as well, PETs can provide an interface with bi-directional power flow capabilities for charging and discharging storage elements.

1.1.1 Power Electronic Transformers for harnessing wind energy

PETs are especially attractive in wind energy generation systems [6]. In conventional on-shore wind turbines, the low-frequency (60Hz) transformers are situated at the base



Figure 1.2: Conventional wind turbine systems



Figure 1.3: Wind turbine systems with low-frequency transformer in the nacelle

of the turbine as shown in Fig. 1.2. Bulky cables carry high currents (kilo-amperes) from the nacelle to the foot of the tower, these cables are costly and also result in high conduction losses. The low-frequency transformers weigh approximately 7-10 tons (2MW, 60Hz). Recently, in off-shore wind turbines, the bulky low-frequency transformers are situated in the nacelle itself as shown in Fig. 1.3 (Vestas V90), this leads to increased cost of infrastructure. These low-frequency transformers can be replaced by high-frequency transformers with the help of power electronics as shown in Fig. 1.4. This would translate to a reduction not only in the cost of infrastructure and copper but also power loss.



Figure 1.4: Proposed Power Electronic Transformer for wind turbine



Figure 1.5: First Power Electronic Transformer [1]

1.1.2 Power Electronic Transformers in PHEVs

Rising gasoline prices and a growing concern about pollution due to the use of fossil fuels have led to the development of Hybrid Vehicles, Electric Vehicles and Plug-in Hybrid Electric Vehicles (PHEVs). PHEVs have the ability to charge the on-board battery pack from the grid or an external power source. Vehicle-to-grid (V2G) is a key storage technology proposed for PHEVs wherein a bi-directional AC-DC converter can be used to provide charging and discharging functionality between the battery and the grid [7]. A PHEV with V2G capability can be charged during off-peak hours at night when the grid is under utilized. It can then be connected to the grid as a distributed resource to provide storage that can be used as standby power for a home, spinning reserves for the grid or for peak load shaving [8–10].

1.2 Three-phase AC-AC Power Electronic Transformers

AC-AC PETs that have either medium frequency transformers or high-frequency transformers are divided into two groups, depending on the input and output frequency, f_i and f_o respectively. The first group has applications in replacing distribution transformers with power electronic transformers where the input and output voltages are at the same frequency [2]. The advantages of doing this are in reducing the size of the transformer, while at the same time obtaining voltage regulation, power flow control, fault current limitation and other features [11–13]. The PET proposed by [1] is shown in Fig. 1.5. In this single-stage power converter, the input voltage is chopped to high frequency using a primary side static converter and a secondary side converter is then used to convert these high frequency voltages to variable magnitude output voltages.



Figure 1.6: Multi-stage Power Electronic Transformer

Voltage regulation is achieved by introducing a phase shift between the input and output converter pulses. Other single-stage Power Electronic Transformers (PETs) have been proposed as distribution transformers in [4,11,14–18]. As the switches have finite turn-on and off times, during any switching transition, the current flowing through the inductive leakage inductance of the transformer requires a path to flow. Hence, this converter requires no storage except during switching intervals when the leakage energy of the transformer flows through a protective clamp circuit.

The second group of PETs are proposed for variable frequency drive applications. PETs used as adjustable speed drives make efficient systems because of their inherent regenerative capabilities [19]. These converters are highly suitable for applications in ships, submarines and aircrafts. Such PETs may also find applications in the microgrids where power flow can be controlled even with a frequency disturbance in the micro-grid [20].

The primary switches in [1] are required to block the entire primary voltage and the secondary switches are required to be rated for the secondary side load currents. This will increase the switch stress as well as cost of the power electronic devices. Hence, multi-level multi-stage PETs have been proposed in [21–26]. Typically, the first stage provides power factor correction and rectification. The following DC-DC converter stage provides high frequency isolation and voltage transformation. The last stage converts the DC voltage to low frequency AC. One drawback is that these converters use DC-link capacitors that are unreliable under thermal stress and hence reduce the overall reliability of the system. Besides, advances in high voltage/high current power electronic devices such as SiC with good switching characteristics, low conduction loss and ability to perform at high temperatures have refocused the interest in single stage power conversion for variable frequency AC-AC converters [5].

Matrix Converters (MC) are a more silicon solution to direct AC-AC conversion. Matrix converters have inherent properties of 1) bi-directional power flow 2) variable frequency and voltage generation 3) active and reactive power control 4) single-stage power conversion. These properties make matrix converters suitable for application in PETs. PETs based on direct modulation [27–29] and indirect modulation [30–32] of matrix converters have been proposed. However, they have a number of switches on the high-voltage side. State-of the art MC based PETs are discussed in Chapter 2.

A High Frequency Transformer (HFT) linked three-phase AC-AC system with only two switches on the high-voltage side was proposed in [6,33–37]. This topology combines the advantages of both matrix converters as well as high frequency transformers. The advantages of this converter are 1) single-stage power conversion, 2) galvanic isolation, 3) power factor control 4) 0.5 modulation index with least number of switching transitions and ZCS for the primary switches. This configuration will be the focus of the first half of this thesis.

1.3 Three-phase AC-DC Power Electronic Transformer

Single-stage isolated AC-DC converters that use cycloconverters are proposed in [38–40]. They have bi-directional power flow capabilities making them suitable for motor drive application for regenerative braking and battery charging/discharging in PHEVs. Protective clamp circuits will be required for these converters because they have finite transformer leakage inductance. This leakage energy commutation limits the switching frequency of these converters [38]. The switching frequency is also limited by the device stress, and switching losses. Converters having soft-switching characteristics, [11, 41–43](multistage) and [44–48](single-stage) are being explored so that they can be operated at higher frequency consequently reducing the size of the transformer and filter.

AC-DC converters with bi-directional capabilities have been proposed in applications for battery charging/discharging in plug-in hybrid vehicles [48], propulsion systems, UPS systems [49], DC grid in ships [50] just to name a few. Bi-directional power transfer between two AC voltages connected by an inductor can be achieved by introducing a phase shift between the two voltages. In Dual Active Bridge (DAB) converters, this phase shift is introduced at high-frequency. This is done either in the DC-DC converter stage or in single-stage AC-DC converters. An advantage being, the transformer leakage inductance is used for power transfer and clamp circuits are no longer required for commutation of the transformer leakage inductance currents. Additionally, these converters have soft-switching characteristics under certain operating conditions [51, 52]. More details on DAB based power converters is covered in Chapter 4. A discussion of current state-of-the-art soft-switched three-phase AC-DC PETs is given in Chapter 5.

In the second half of this thesis, a single-phase and three-phase bi-directional, HFT isolated, single-stage, DAB-based, AC-DC converter are analyzed. These converters have only two controlled switches on the AC-side. The transformers provide isolation and their leakage inductance is used for power transfer. This topology combines all the advantages of a HFT based system and a DAB-based system. Ideally, no clamp circuit is required for this converter.

1.4 Contributions of this thesis

The two main contributions of this thesis are analysis, design and implementation of laboratory prototypes of 1) a novel AC-AC Matrix converter based PET 2) a novel single-phase and three-phase AC-DC PET based on DAB principle. Both these converters employ high-frequency transformers in push-pull configuration. They have the same primary circuit with the purpose of converting three-phase line voltages to highfrequency AC voltages. The secondary side converter transforms these high-frequency voltages to either to three-phase variable magnitude and frequency AC voltages or DC voltage. These converters have features of single-stage power conversion, bi-directional power flow capabilities and power factor correction.

The problems associated with the finite transformer leakage inductance of the AC-AC PET have been thoroughly studied. A modulation strategy is proposed that eliminates the primary clamp circuitry for the primary side converter. The single-phase and three-phase AC-DC converters proposed in this research use the transformer leakage inductance to enable power transfer, hence they do not suffer from the drawbacks of increased loss and voltage distortions associated with clamp circuits.

1.5 Organization of this thesis

This thesis is organized into four main chapters, the first two focus on a PET for AC-AC conversion and the next two on PET for AC-DC conversion. A brief description of each chapter is given below.

- Chapter 1 introduces the goals pursued in this thesis.
- Chapter 2 introduces a PET with reduced number of switches for AC-AC power conversion. This chapter presents an analysis of the protective snubber circuits (clamp circuits) requirements on the primary side. A new modulation technique is proposed in order to reduce the protection requirements in the source side. This topology with the proposed modulation technique is analyzed and simulated. The experimental results of this converter are presented.
- Chapter 3 provides a detailed analysis of the power loss incurred in the clamp circuit along with a method to design the clamp components. The entire circuit is simulated along with non-ideal leakage inductance and the presented simulation results are compared with the analytical predictions. The effects of dead-time commutation and four-step commutation are compared in this chapter. The experimental results are presented.
- Chapter 4 proposes a control method for a single-phase AC-DC PET based on Dual Active Bridge principle. The converter is thoroughly analyzed by first assuming it to be a DC-DC converter in a push-pull topology and then extending the results to analyze an AC-DC converter. The conclusions of the analysis are confirmed by simulations. The experimental results for this converter are presented.
- Chapter 5 proposes a modulation technique based on the Dual Active Bridge principle for a single-stage three-phase AC-DC PET with only two active switches on the AC-side. This converter is analyzed in detail. The experimental results are presented.
- Chapter 6 summarizes and concludes the thesis.

Chapter 2

AC-AC Power Electronic Transformer: Topology and Modulation

2.1 Introduction

Matrix converter based three-phase AC-AC power electronic converters have inherent bi-directional energy flow capability, power factor correction and single stage power conversion without any storage elements (capacitors and inductors are required only for filtering and protection). High-frequency transformers (HFT) provide galvanic isolation along with voltage transformation. A major advantage being their small size as compared to line frequency transformers. HFT linked matrix converter based three-phase AC-AC systems, combine the advantages of the matrix converter and the HFT [53]. A summary of the state-of-the-art matrix converter based Power Electronic Transformers is given in this introduction.

A PET based on indirect modulation of the matrix converter proposed by [30] uses two three-phase to single-phase matrix converters linked by a high-frequency transformer. Hence, this topology has 12 switches on the high-voltage side. A PET that uses a three winding transformer instead of the two winding transformer of [30] is proposed in [31]. The operating principle for the primary converter is the same as that of [30], however, due to availability of three levels on the secondary side, common-mode voltage elimination can be achieved in this converter. Another PET, based on indirect modulation of matrix converter that has four wire outputs that is better suited for operation under unbalanced operation is proposed in [32]. A PET based on direct modulation of MC is presented in [27]. It uses two MC on the primary side, has a voltage transformation ratio of up to $1.5 \times \sqrt{3}/2$ and has zero common-mode voltages in the output voltages. However, it has 36 switches on the high voltage side and $18 \times 3 = 54$ switches in all.

The High-frequency transformer (HFT) link in PETs have finite leakage inductance, hence any change in the switching state of the input or output side converter of the PET needs commutation of leakage energy. Commutation of leakage energy using clamp circuits leads to power loss, voltage distortion and requires higher voltage blocking capabilities for the devices in the converter [54]. Some of the problems associated with the commutation of leakage energy can be solved by source-based commutation [29]. Source based commutation is a technique where the primary switches are modulated in a smart way such that the the input voltages instead of clamp circuits are used for commutation purposes. A topology that uses lossless source based commutation technique is proposed in [28]. It uses two matrix converters on the primary side and six bi-directional switches on the secondary side. Another variation of [30] has been proposed in which source based commutation is possible [55]. Although source based commutation is lossless, these PETs have many switches on the high-voltage side and they use very precise and complicated switching strategies.

The High-Frequency Transformer linked AC-AC system shown in Fig. 2.1 with only two switches on the high-voltage (primary) side was proposed and analyzed in [6,33–37]. Source based commutation is not possible in this topology, hence, a clamp or a snubber circuit is used on both sides of the transformer for safe commutation of leakage currents.

This chapter presents a comprehensive analysis of the primary (input) clamp circuit of the proposed PET in Fig. 2.1 and the resulting limitations. A modulation strategy for the secondary (output) side matrix converter has been proposed in order to obviate the need for an input clamp circuit. The advantages of the PET and the proposed modulation strategy are 1) single-stage power conversion, 2) galvanic isolation, 3) power factor control 4) 0.5 modulation index with least number of switching transitions and



Figure 2.1: Proposed AC-AC PET topology

Zero Current Switching (ZCS) for the primary switches. Experimental results on a laboratory prototype confirm the advantages of the proposed modulation strategy.

2.2 PET topology and modulation strategy

In this topology, the input three-phase balanced AC voltages are applied to a bank of three single-phase transformers as shown in Fig. 2.1. S_1 and S_2 on the primary side are switched in a complimentary fashion with 50% duty cycle. When switch S_1 is on, terminals a_1 , b_1 and c_1 are shorted and the upper half of the primary windings conduct. Similarly, when S_2 is on, terminals a_2 , b_2 and c_2 are shorted and power is transferred through the lower halves of the primary windings. The voltage across each secondary winding is a chopped version of the corresponding input line to neutral voltage. In the secondary side of the transformers, a matrix converter is employed in order to generate adjustable magnitude and frequency AC waveforms. The Matrix converter is modulated using rotating vectors as described in [33] and [6] to obtain a modulation index of upto 0.5 and 0.866 respectively. During each switching transition of S_1 and S_2 (T_1 and T_2 as shown in Fig. 2.2), power flow is transferred from one half of the primary windings to the other. Due to the presence of leakage inductances L_1 and L_2 (Fig. 2.3) in the primary windings of the transformer, this transition is not instantaneous, and a snubber or clamp circuit is required for the commutation of the primary leakage energy. In Fig. 2.3, diodes d_{11} , d_{22} and the clamp capacitor C_l form the required clamp circuit. As both the transitions $(T_1 \text{ and } T_2)$ of S_1 and S_2 are symmetrical, here, a detailed analysis of T_1 is presented. During T_1 and T_2 , the output matrix converter is not switched. The secondary side matrix converter with three-phase balanced load can be modeled as three-phase balanced current sources $(i_A = I_A, i_B = I_B \text{ and } i_C = I_C)$. Without any loss of generality, it is assumed that I_A is positive, while I_B and I_C are negative. Just before T_1 , the currents i_{a_1} , i_{b_1} and i_{c_1} are equal to I_A , I_B and I_C respectively and i_{a_2} , i_{b_2} and i_{c_2} are all zero. The diodes, d_{a_1} , d_{b_2} and d_{c_2} are conducting to provide a path for these currents to flow through S_1 . At T_1 , when S_1 is switched off, the currents i_{a_1} , i_{b_1} and i_{c_1} cannot change instantaneously, this forces diode d_{11} to come into conduction and current starts flowing through the capacitor C_l . In order to simplify the analysis, it is assumed that the two halves of the primary winding and the secondary winding have equal number of turns and $L_1 = L_2 = L$. Neglecting the magnetizing current, application of Amperes' law results in (2.1). In Fig. 2.3, two conducting loops can be traced involving upper halves of the primary windings. The KVL equations for these two loops are given by (2.2) and (2.3). By (2.1), if i_{a_1} reduces, the diode d'_{a_2} will turn on and the lower winding of phase-a will start conducting. Similarly, for a desired change in currents i_{b_1} and i_{c_1} , diodes, d'_{b_1} and d'_{c_1} will start conducting. Switch S_2 provides a path for these currents to flow. The KVL equations for the two conducting loops with the lower halves of the primary windings are give by (2.4) and (2.5). In a balanced three-phase system, with a floating neutral point, the three line to neutral values must sum to zero hence we obtain (2.6).

$$i_{a_1} - i_{a_2} - i_A = 0$$

$$i_{b_1} - i_{b_2} - i_B = 0$$

$$i_{c_1} - i_{c_2} - i_C = 0$$
(2.1)



Figure 2.2: Switching pulses for S_1 and S_2



Figure 2.3: Primary clamp circuit analysis

14

$$v_a - v_{a_1} - L\frac{d}{dt}i_{a_1} - V_{clp} + L\frac{d}{dt}i_{c_1} + v_{c_1} - v_c = 0$$
(2.2)

$$v_a - v_{a_1} - L\frac{d}{dt}i_{a_1} - V_{clp} + L\frac{d}{dt}i_{b_1} + v_{b_1} - v_b = 0$$
(2.3)

$$v_c + v_{c_1} - L\frac{d}{dt}i_{c_2} + L\frac{d}{dt}i_{a_2} - v_{a_1} - v_a = 0$$
(2.4)

$$v_b + v_{b_1} - L\frac{d}{dt}i_{b_2} + L\frac{d}{dt}i_{a_2} - v_{a_1} - v_a = 0$$
(2.5)

$$v_{a_1} + v_{b_1} + v_{c_1} = 0 (2.6)$$

There are nine equations (2.1) to (2.6), involving six unknown rates of change of primary current and three unknown induced voltages in the transformer windings. The slope of the primary side currents are given by (2.7)- (2.9).

$$\frac{d}{dt}i_{a_1} = \frac{d}{dt}i_{a_2} = \frac{3v_a - V_{clp}}{3L}$$
(2.7)

$$\frac{d}{dt}i_{b_1} = \frac{d}{dt}i_{b_2} = \frac{6v_b + V_{clp}}{6L}$$
(2.8)

$$\frac{d}{dt}i_{c_1} = \frac{d}{dt}i_{c_2} = \frac{6v_c + V_{clp}}{6L}$$
(2.9)

When the switch S_1 is switched off and S_2 is switched on, the currents in the transformer are changing according to equations (2.7)-(2.9). When one of the three currents reaches zero, in the winding set connected to S_1 and the corresponding winding current connected to S_2 gets set to its desired value, the currents in the remaining phases, continue to flow at different rates. In the following analysis, let us assume that i_{b_1} goes to zero and i_{b_2} becomes I_B while, i_{a_1} and i_{c_1} are still positive and negative respectively. The network is changed because the branch containing diodes d_{b_1} and d_{b_2} has no current flowing through them and can be considered as an open circuit. All the equations remain the same as in the previous analysis except equation (2.3) is no longer valid, and the currents i_{b_1} and i_{b_2} do not change. In this stage, the slope of the currents, are given by (2.10)-(2.12).

$$\frac{d}{dt}i_{a_1} = \frac{d}{dt}i_{a_2} = \frac{2v_a - 2v_c - V_{clp}}{4L}$$
(2.10)

$$\frac{d}{dt}i_{b_1} = \frac{d}{dt}i_{b_2} = 0 \tag{2.11}$$

$$\frac{d}{dt}i_{c_1} = \frac{d}{dt}i_{c_2} = -\frac{2v_a - 2v_c - V_{clp}}{4L}$$
(2.12)

If the voltage across the clamp circuit capacitance (V_{clp}) is greater than 6 times the peak of the input line-neutral voltage (V_{ln_i}) , all the primary currents during commutation will change in the desired direction. For example, in this case, i_{a_1} being positive, before T_1 , should reduce to zero during commutation. Similarly, i_{a_2} should reduce to $-I_A$. This condition also ensures that once the primary currents reach their desired values, the commutation ends naturally. For example, once i_{a_1} reaches zero it remains there. The primary snubber circuit is operational only two times during one switching period, the average current that flows through the clamp circuit per cycle is equal to the sum of the average value of i_{a_1} during T_1 . For this case, the average clamp power loss is given by (2.13).

$$P_{pri} = \frac{1}{2} f_s L V_{clp} \left[\frac{3I_B^2}{6v_b + V_{clp}} - \frac{(I_A - I_C)^2}{2v_a - 2v_c - V_{clp}} \right]$$
(2.13)

From this analysis, the following points are evident 1) for the proper operation of the clamp circuit, V_{clp} has to be maintained greater than six times the line to neutral voltage. 2) From Fig. 2.3, when d_{11} conducts, the switch, S_1 has to block the voltage across the clamp circuit. 3) During the commutation, undesirable voltages are applied to the load, this leads to distortion in the output voltage and eventually in the output currents. 4) This commutation process also results in power loss given by (2.13). A new modulation method is presented in the following section that overcomes these drawbacks.

2.3 A PWM technique to eliminate the input clamp circuit

The output side matrix converter synthesizes the high-frequency output waveform of the transformer into adjustable frequency and amplitude pulse width modulated (PWM) voltages at terminals u, v and w. When a zero vector is applied in the matrix converter, the secondary currents i_A , i_B and i_C become zero with the help of the secondary side clamp circuit. Then, S_1 and S_2 can be safely switched without the primary side snubber circuit. The input line to neutral voltages and the secondary side voltages are given by

(2.14) and (2.15) respectively; where, k = 0 when S_1 is on and k = 1 when S_2 is on.

$$v_{an_i} = V_i \cos(\omega_i t)$$

$$v_{bn_i} = V_i \cos\left(\omega_i t - \frac{2\pi}{3}\right)$$

$$v_{cn_i} = V_i \cos\left(\omega_i t + \frac{2\pi}{3}\right)$$
(2.14)

$$v_{AB} = (-1)^{k} v_{ab}
 v_{BC} = (-1)^{k} v_{bc}
 v_{CA} = (-1)^{k} v_{ca}$$
(2.15)

In a matrix converter, there are 27 unique switching states. Six of these switching states generate synchronously rotating space vectors [56]. These vectors result in zero common-mode voltage at the load terminals [57]. The output voltage space vector is defined by (2.16). Depending on the direction of rotation, these synchronously rotating vectors are further divided into two groups, counter-clockwise (CCW) and clockwise (CW). The three CCW rotating vectors $\vec{V_1}$, $\vec{V_3}$ and $\vec{V_5}$ marked in Fig. 2.4 are obtained when k = 0 and terminals $\{u, v, w\}$ are connected to terminals $\{A, B, C\}, \{C, A, B\}$ and $\{B, C, A\}$ respectively which are marked in Fig. 2.1. For the same switching states, when k = 1, we obtain \vec{V}_4 , \vec{V}_6 and \vec{V}_2 respectively. Similarly, the space vectors rotating in clockwise direction are shown in Fig. 2.5. Thus, every switching cycle, the average output voltage vector is synthesized using these six vectors (CCW or CW). These six vectors, divide the complex plane into six symmetrical sectors (marked in Fig. 2.4 and 2.5). At any instant of time, the reference voltage vector (defined by (2.17)) will be in a particular sector. This reference voltage vector is generated on an average over one T_s using the two vectors forming that sector. For example, if \overline{V}_{ref} lies in the first sector, vectors \vec{V}_1 and \vec{V}_2 are used, such that $\overline{V}_{ref} = d_1 \vec{V}_1 + d_2 \vec{V}_2$, where d_1 and d_2 (given by (2.18)) are the fraction of time for which vectors \vec{V}_1 and \vec{V}_2 are applied respectively. The values of d_1 and d_2 are constrained to 0.5 because \vec{V}_1 and \vec{V}_2 are available for maximum 50% of the time. Hence, if only the two adjacent space vectors are used, the maximum modulation index, m is limited to 0.5. In [6], three adjacent voltage vectors are used to



Figure 2.4: CCW rotating vectors

Figure 2.5: CW rotating vectors

obtain a modulation index of 0.75 at the cost of additional switching.

$$\vec{V}_{o} = v_{un_{o}} + v_{vn_{o}}e^{j\frac{2\pi}{3}} + v_{wn_{o}}e^{-j\frac{2\pi}{3}}$$

$$\overline{V}_{ref} = \overline{V}_{o} = \overline{v}_{un_{o}} + \overline{v}_{vn_{o}}e^{j\frac{2\pi}{3}} + \overline{v}_{wn_{o}}e^{-j\frac{2\pi}{3}}$$

$$\overline{v}_{un_{o}} = V_{o}\cos\left(\omega_{o}t + \phi\right)$$

$$\overline{v}_{vn_{o}} = V_{o}\cos\left(\omega_{o}t + \phi - \frac{2\pi}{3}\right)$$

$$\overline{v}_{wn_{o}} = V_{o}\cos\left(\omega_{o}t + \phi + \frac{2\pi}{3}\right)$$

$$(2.16)$$

The switching pulses for one cycle are shown in Fig. 2.6. The duty ratios d_1 and d_2 are compared with a triangular carrier waveform, V_{tri} to generate pulses p_{V_1} and p_{V_2} . V_{tri} has a peak value of 0.5 and its frequency is two times the frequency of S_1 and S_2 . In sector one, $\vec{V_1}$ is available when S_1 is on, therefore, p_{V_1} is high in the first half of the cycle and p_{V_2} in the second half. Zero vectors are applied in the remaining time. p_z is the switching pulse for the zero vector. In each half cycle, the active vectors are buffered by zero vectors on either side.

Ignoring the effects of the input filter, if only CCW vectors are used for modulation, the grid power factor will equal the load power factor. If only CW vectors are used for modulation, the power angle of the PET will have the same magnitude as of the load



Figure 2.6: Switching pulses for new modulation technique

V_i, V_o, V_{clp}	100V, 30V, 600 V	
ω_i, ω_o	$2\pi 60 \text{ rad/sec}, 2\pi 60 \text{ rad/sec}$	
Output power	1kW	
Load pf.	0.8	
L	$12 \mu H$	

Table 2.1: Simulation parameters

but of opposite sign. If CCW and CW rotating vectors are used for an equal duration of time, unity power factor is obtained on the primary side of the PET [58]. The input power factor can further be varied by changing the reference voltage vector for CCW and CW vectors [57].

$$d_{1} = m \frac{2}{\sqrt{3}} \sin\left(\frac{\pi}{3} - \alpha\right)$$

$$d_{2} = m \frac{2}{\sqrt{3}} \sin(\alpha)$$

$$m = \frac{V_{o}}{V_{i}}$$
(2.18)



Figure 2.7: Simulation result: input clamp circuit analysis

2.4 Simulation results

The circuit in Fig. 2.3 is simulated in SABER®. The leakage inductance value is chosen to be 12μ H, and V_{clp} is kept at 700V. The instantaneous line to neutral voltages are set as follows $v_{an_i} = 100V$, $v_{bn_i} = -40V$ and $v_{cn_i} = -60V$. The secondary side current sources I_A , I_B and I_C are set to 20A, -5A and -15A respectively. In Fig. 2.7 switch S_1 is turned off at 50 μ s (transition T_1). The currents i_{a_1} , i_{b_1} and i_{c_1} go to zero following the analytically predicted slopes. The current that flows into the clamp circuit is $i_{d_{11}}$, as expected it is the same as i_{a_1} .

A 1kW three-phase AC-AC PET is simulated with the proposed modulation technique. The parameters used in the simulation are listed in Table 2.1. Fig. 2.8(a) shows the input line to neutral voltage waveform with the corresponding filtered line current. This confirms input power factor correction. Fig. 2.8 (b) provides the sinusoidal output load current. The peak of this current is slightly lower than its analytically predicted value. This is due to the voltage loss resulting from the use of a secondary clamp circuit. From Fig. 2.9, it is observed that the currents through the primary side switches are zero when S_1 and S_2 are switched.

2.5 Experimental setup

The experimental setup is shown in Fig. 2.11. Schematically it is represented in Fig. 2.10. All the power electronic devices are controlled using a single Xilinx XC3S500E FPGA. The primary switches, S_1 and S_2 are given pulses at 50% duty at 10kHz. The PWM


Figure 2.8: Simulation result: Three-phase AC-AC HFT using proposed modulation method (a) input voltage (1V/div) and input current (0.1A/div) (b) output current (1A/div)



Figure 2.9: Simulation result: Switching pulse and currents through switch S_1 and S_2

Table 2.2: Experimental setup: parameters

V_i, V_o	$80\sqrt{2}V (SVM); 40\sqrt{2}V(CBM), 44.46V(SVM); 22.23V (CBM)$
$\omega_i \; , \omega_o \; , f_s$	$2\pi 60, 2\pi 25$ (SVM); $2\pi 50$ (CBM), 10kHz
R , L	5.5Ω , $30\mathrm{mH}$
Transformer turns ratio	1:1:1
$L_{filter}, C_{filter}, R_{filter}$	$0.5 \mathrm{mH}, 20 \mu \mathrm{F(star)}, 2 \Omega$

pulses for all the switches are generated after the input voltages v_{ab} and v_{ac} are sensed. In order to perform four-step commutation in the MC, two other ADCs sense the output currents i_u and i_v . When the current direction is accurately known, four-step commutation is performed else dead-time commutation is performed. A comparison of four-step and dead-time commutation for this topology is in Chapter 3.

In this chapter, the experimental results are provided for modulation of the matrix converter using Carrier Based Modulation (CBM) [58] and Space Vector Modulation (SVM) [35] that is proposed in this chapter. The experimental parameters for these two modulation strategies are in Table 2.2.

The details of the hardware setup are in Appendix B. The details on FPGA implementation for carrier based modulation and space vector modulation are in Appendix D.

The high-frequency transformers are wound on ferrite cores. These transformers are designed according to the area-product method and have leakage inductance in the range of 15μ H and magnetizing inductance of 50mH. The details of the transformer design are in the Appendix C.

The input filter is designed to filter out the high-frequency switching currents at the input of the PET. It is composed of $L_f = 0.5$ mH and $C_f = 20\mu$ F with a resistance of $R_f = 2\Omega$ in parallel with L_f to damp out the LC resonant frequency of the filter.

2.5.1 Transformer voltages

The primary and secondary voltages of the high-frequency transformer, v_{ab} and v_{AB} respectively are shown in Fig. 2.12. It can be seen that the 60Hz low-frequency input voltages are chopped to high-frequency AC voltages at 10kHz. During the dead-time (t_{dt}) between the switch S_1 and S_2 , the primary clamp circuit voltage appears across



Figure 2.10: Experimental setup



Figure 2.11: Experimental setup for PET for AC to AC power conversion



Figure 2.12: Transformer primary v_{ab} (C3: 100V/div) and secondary voltage v_{AB} (C4:100V/div) (top) along with their zoomed waveform at 100 μ s/div (bottom).

the transformer windings and a spike in voltage is observed. This clamp circuit is a passive RC circuit that needs to be maintained at least two times the line to neutral voltage, during the dead-time, the clamp circuit draws power from the input to support this voltage.

2.6 Experimental results: Carrier based modulation

Carrier based modulation is implemented for the following three cases 1) Using only CCW rotating vectors 2) using only CW vectors and 3) using CCW and CW vectors for equal duration of time. The results shown in Fig. 2.13 are taken at $V_i = 40\sqrt{2}$ V, $V_o = 22.23V$ with $\omega_i = 2\pi 60$, $\omega_o = 2\pi 50$, $f_s = 10kHz$. The *RL* load for all these cases is the same and equals 5.5 Ω and 30mH. The output current for all these cases is 1.19A. The experimental results are summarized in Table. 2.3. It can be seen that the input power factor can be varied by changing the vectors used for modulation.

The transformer currents for phase-*a* are in Fig. 2.14. The upper winding conducts when S_1 is high and the lower winding conducts when S_1 is off. The currents have three distinct sections that correspond to the three vectors applied to the matrix converter. The currents are not zero at the transition of S_1 hence the primary side switches are not ZCS.



(c) CCW+CW vectors

Figure 2.13: Experimental results for carrier based modulation: (top) Filtered input current i_{af} (C1:1A/div) and voltage v_{an} (C3:20V/div) for phase-*a*; (bottom) output current i_u (C2:1A/div) and voltage v_{un_o} (50V/div) for phase-*u* when (a) CCW, (b) CW and (c) CCW and CW vectors are used for equal duration of time.



Figure 2.14: (left) Transformer currents for phase- $a i_{a1}$, i_{a2} , i_A (C1,C2, C4 : 2A/div) and switching pulse for S_1 ; (right) zoomed waveforms at (50 μ s/div).

2.7 Experimental results: Space vector modulation

The modulation for the PET is done for three cases 1) using CCW rotating vectors only, 2) using CW vectors only and 3) using CCW and CW vectors for equal duration of time. The experimental parameters of the PET are listed in Table 2.2. In all three cases, the output voltages at 25Hz are supplied to an RL load of 5.5 Ω and 30mH. As the input voltages and load are balanced-three-phase, it is sufficient to analyze the results of a single phase only. The waveforms for filtered current (i_{af}) , phase-*a* voltage (v_{an_i}) as well as output current (i_u) and output voltage (v_{un_o}) for these three cases are in Fig. 2.15. In all these cases, the output current i_u has an RMS value of 4.11A. But the input current depends on the vectors used for modulation.

rabie 2.6. Experimental results. Carrier sased modalation						
	$I_{af}(\mathbf{A})\mathbf{RMS}$	$P_{in}(Watt)$	$Q_{in}(VA)$	$\theta(\text{deg})$		
CCW	0.44	17.92	3.58	-11.29		
CW	0.74	17.72	-24.92	54.59		
CCW+CW	0.53	18.10	-12.95	35.57		

Table 2.3: Experimental results: Carrier based modulation



(c) CCW+CW vectors

50.0 Vidi

50.0 Vidir

Figure 2.15: Experimental results for space vector modulation: (top) Input current i_{af} (C1:2A/div) and voltage v_{an} (C3:50V/div) for phase-*a*; (bottom) output current i_u (C2:2A/div) and voltage v_{un_o} (50V/div) for phase *u*. (a) CCW, (b) CW and (c) CCW and CW vectors are used for equal duration of time.



Figure 2.16: Real and reactive power of the PET

2.7.1 Power factor control

The reactive power drawn from the source can be divided into two parts; Power to the input filter (Q_{filter}) and power of the matrix converter. As the transformer magnetizing currents are balanced at high-frequency of 10kHz, it does not contribute to the reactive power at the fundamental of 60Hz. When CCW vectors are used, the input reactive power is the same as the load reactive power in addition to Q_{filter} (2.19). When CW vectors are used, in addition to Q_{filter} , the input reactive power is equal in magnitude to the load reactive power but opposite in sign (2.20). For these cases, at a certain input voltage, and output load, the voltage drop across the filter inductor L_f is approximately the same and very small. Hence, it can be assumed that $Q_{filter} = Q_0$. When CCW and CW vectors are used for equal duration of time, the net input reactive power is Q_{filter} (2.21). The power diagram for the active and reactive power of this PET with an inductive load of $Q_L = V_o I_o \sin(\rho)$ where ρ is the load angle is in Fig. 2.16. In this experiment, $Q_L = 79.6$ VA and $Q_0 = -44.8$ VA. The active power consumed by the PET remains the same in all three cases, but the reactive power changes as shown in Table 2.4.

$$Q_{ccw} = Q_{filter} + Q_L \tag{2.19}$$

$$Q_{cw} = Q_{filter} - Q_L \tag{2.20}$$

$$\frac{Q_{ccw} + Q_{cw}}{2} = Q_0 \tag{2.21}$$

27

	I_{af} (A)RMS	P_{in} (Watt)	Q_{in} (VA)	θ (deg)
CCW	1.59	123.92	41.55	-18.54
CW	2.17	122.59	-131.17	46.94
CCW+CW	1.66	124.11	-58.62	25.28

Table 2.4: Power factor control

$$\cos(\theta_{ccw}) = \frac{P}{\sqrt{P^2 + (Q_0 + Q_L)^2}}$$
(2.22)

$$\cos(\theta_{cw}) = \frac{P}{\sqrt{P^2 + (Q_0 - Q_L)^2}}$$
(2.23)

$$\cos(\theta_{ccw+cw}) = \frac{P}{\sqrt{P^2 + Q_0^2}}$$
 (2.24)

The input power factor for the three cases is given by (2.22)- (2.24). The input side power angle is most leading when only CW (θ_{cw} analytical 45.42⁰, observed 46.94⁰) vectors are used for modulation. It is most lagging when only CCW vectors (θ_{ccw} analytical -15.85° , observed -18.54°) are used for modulation. It is concluded that by use of CCW and CW vectors in varying proportions, the reactive power drawn by the PET can be varied up to a value that depends on the load reactive power. The slight discrepancy is due to the simplifying assumption that the reactive power drawn by the filter inductor is approximately the same for all three cases.

2.7.2 Fourier analysis of input currents

The Fourier spectrum of the input current of the PET for phase-*a* as shown in Fig. 2.17 are plotted by running an FFT with a hann window on the discrete current data that is collected from the scope. The signal is first filtered and decimated to remove any high-frequency components beyond 80kHz and reduce the sample rate for easier processing.

When either CCW vector or CW vectors are employed, the input current has harmonics at the fundamental as well as at the switching frequency of 10kHz as shown in Fig. 2.17(a) and Fig. 2.17(b). When CCW+CW vectors are used, a 5kHz component is observed in the input current waveform as shown in Fig. 2.17(c). The CCW and CW vectors are used for a duration of 100μ s each, hence, the average current vector is synthesized over 200μ s. The fundamental component of i_a at 60Hz when only CCW



Figure 2.17: Fourier spectrum of input current for phase-a

and only CW vectors are used is 1.48A RMS and when CCW and CW vectors are used for equal duration of time, the input current is scaled by $\cos(\rho)$ and equals 1.25A RMS.

In order to calculate the Total Waveform Distortion (TWD) given by (2.25), the fundamental component is calculated by running a DTFT at the fundamental frequency to calculate I_{f1} . The value of I_{rms} is calculated from the discrete data. The TWD for the filtered input currents when CCW vectors are applied is 7.85%, when CW vectors are applied it is 8.37% and when CCW+CW vectors are used it is 13.94%.

$$TWD = \frac{\sqrt{I_{rms}^2 - I_{f1}^2}}{I_{f1}} \tag{2.25}$$

2.7.3 Soft-switching of primary converter

The currents through transformer winding 1 and 2 for phase- a (i_{a1}, i_{a2}) and the corresponding secondary side current (i_A) are shown in Fig. 2.18(a). The currents through the secondary windings of the transformers, i_A , i_B and i_C are shown in Fig. 2.18(b). It can be seen that when a zero vector is applied in the MC, the transformer secondary currents become zero. Hence, ZCS is achieved in S_1 and S_2 .

There is a spike of current during the dead-time $(1.5\mu\text{sec})$ between the transitions of S_1 and S_2 , power is supplied to the primary clamp circuit which is there for protection purposes only and is not activated during normal operations.

2.7.4 Output current and voltage

The output currents for phase u and v as well as the line to line voltages are shown in Fig. 2.19. The input current in Fig. 2.19(a), 2.19(c) and 2.19(e) has additional ripple because the input LC filter is not damped. There are spikes in the output lineline voltages when four-step commutation is suspended and dead-time commutation is implemented.

2.7.5 Extended power factor control

With the use of a combination of CW or CCW vectors, the power factor control of the PET is limited by the load power angle (ρ). Hence, a method proposed in [57] for open-end winding drives is applied in this PET where power factor control out of this



Figure 2.18: Left: Transformer currents for phase-*a* i_{a1} , i_{a2} , i_A (C1,C2,C4:2A/div) and switching pulses for S_1 (C3: 10V/div); Right: transformer secondary currents i_A , i_B , i_C (C1,C2,C4:2A/div) and switching pulses for S_1 (C3:10V/div).

range is possible at the cost of modulation index. This method is briefly described and analyzed for the proposed PET in this section.

Extended power factor control can be demonstrated by the three cases shown in Fig. 2.20, where the vector \vec{V}_{ccw} and \vec{V}_{cw} are generated using CCW and CW vectors respectively and $|\vec{V}_{ccw}| = |\vec{V}_{cw}| = |\vec{V}_o|$. a) CCW and CW vectors generate the same reference voltage of $\vec{V}_o \cos \beta$ for equal duration of time. b) The reference for the CCW vector is $\vec{V}_o \angle -\beta$ and for the CW vectors the reference voltage is $\vec{V}_o \angle \beta$. c) The reference for the CCW vector is $\vec{V}_o \angle \beta$ and for the CW vectors the reference voltage is $\vec{V}_o \angle \beta$. In an average sense, the equivalent output voltage in all three cases is the same and equals $V_o^* = \vec{V}_o \cos \beta$. As the load is constant and inductive in nature with a load angle of ρ . In these cases, the current vector is equal to $I \angle \rho$. The equations for input reactive power for Case b is in (2.26). $Q_{ccw\beta}$ and $Q_{cw\beta}$ are the reactive powers due to the CCW and CW vectors at angle β and $-\beta$ respectively. Q_β is the net reactive power of the PET for Case b. The input reactive power for Case c can be calculated in a similar way.

$$Q_{ccw\beta} = Q_0 + V_o^* I \sin(\rho - \beta)$$

$$Q_{cw\beta} = Q_0 - V_o^* I \sin(\rho + \beta)$$

$$Q_\beta = Q_0 + \frac{1}{2} \left[V_o^* I \sin(\rho - \beta) - V_o^* I \sin(\rho + \beta) \right]$$
(2.26)

The experimental confirmation of variable power factor compensation is proved by



Figure 2.19: Left: (top) Output currents i_u, i_v (C1, C2:2A/div); (bottom) input voltage and current v_{an} (C3:50V/div) and i_{af} (C4:2A/div) for phase-*a*. Right: (top) Output voltages between phase v_{uv} and v_{vw} ; (bottom) input voltage for phase-*a* (50V/div) and output current for phase-*u* at 2A/div.



Figure 2.20: Extended power factor control

	I_{af} (A)RMS	P_{in} (Watt)	Q_{in} (VA)	θ (deg)
Case a	0.58	22.38	-28.56	51.91
Case b	0.78	21.88	-44.04	63.58
Case c	0.43	22.95	-13.83	31.07

Table 2.5: Extended power factor control

running the three cases described above for $V_i = 60\sqrt{2}$, $V_o = 33.35$ V, 50Hz and $\beta = 48^0$. The load is a resistor of 5.5 Ω and inductor of 30mH. The waveforms are in Fig. 2.21 and the results are compiled in Table 2.5. As predicted, the values of $P_{in} = 22$ Watt for these three cases is the same however the values for Q_{in} is different, with Case b (expected -45.06VA observed -44.04VA) having the most leading power factor and Case c (expected -12.81VA observed -13.83VA) having the most lagging power factor.

The experimental values of input reactive power for different values of β are compared with the expected values in Fig. 2.22. It can be seen that they match closely.

2.7.6 Variable frequency generation

The output voltages in Fig. 2.15 are at 25Hz and the output voltages in Fig. 2.21 are at 50Hz. Hence, variable frequency and magnitude output voltages can be generated using this PET.

2.8 Conclusions and future work

In this chapter a novel power electronic transformer is presented that has the following features,

33



(c) Case c

Figure 2.21: Extended power-factor control: (top) Filtered input current i_{af} (C1:1A/div) and voltage for phase-*a* v_{an} (C3:50V/div); (bottom) output current i_u (C2:1A/div)



Figure 2.22: Power factor variation for different values of β

- Single-stage power conversion with galvanic isolation and bi-directional power flow capability.
- Variable frequency and amplitude pulse width modulated voltage generation.
- Power factor control at the input.
- Zero current switching (ZCS) for the input side converter.
- Compact size and easy control technique that can be implemented in a single FPGA.

The simulation/hardware results verify that variable frequency and voltage generation is possible in this PET. CCW and CW rotating vectors can be used in varying proportions to achieve control over the reactive power at the input of this PET. The experimental prototype has been tested extensively and the experimental results match the analytical predictions. The experimental results show that, with the new modulation technique, ZCS is achieved for the primary switches, S_1 and S_2 . This is an improvement over carrier based modulation. Although, the proposed solution eliminates the primary clamp circuit, the secondary clamp circuit is still necessary. The secondary clamp circuit analysis is done in Chapter 3.

2.8.1 Future work

- When CBM is used in the secondary side MC, the common-mode voltages at terminals $\{u, v, w\}$ is nearly zero. When SVM is used, since zero vectors are applied in the matrix converter, switching common-mode voltages are present at terminals $\{u, v, w\}$. Common-mode voltage elimination can be obtained by operating this PET as an open-ended drive.
- The TWD of the filtered currents is high in this PET. Optimal filter design is required to obtain better TWD.

Note: Parts of this chapter have been reprinted from [35] ©2010 IEEE

Chapter 3

AC-AC Power Electronic Transformer: Clamp Circuit Analysis

3.1 Introduction

The Matrix converter based Power Electronic Transformer discussed in Chapter 2 is a single-stage solution with features of bi-directional power flow capabilities, galvanic isolation and power factor control. Theoretically, this PET provides single-stage power conversion and obviates the need for any storage elements. However, any switching transition in the load side converter causes over voltages due to the presence of finite leakage inductances in the windings of the high-frequency transformer. A clamp circuit as shown in Fig. 3.1 is needed for commutation of this leakage energy and to protect the power electronic devices.

This chapter presents a detailed analysis of secondary side leakage energy commutation for the reduced-switch topology in Fig. 3.1 with respect to the modulation method proposed in Chapter 2. The power losses associated with the clamp circuit operation, the commutation time required and the voltage loss incurred are analytically calculated and checked with simulation. The analysis developed in this chapter is quite general and is applicable to analysis of the clamp circuit in other MC based PETs such as topology



Figure 3.1: Topology with non-idealities

B2 and B3 in [53].

Four-step commutation has been proposed for matrix converters where the input is a voltage port and the output a current port [59]. However, in some MC based PETs, the input as well as the output ports have inductance. A comparison of deadtime commutation and four-step commutation for the MC based PET in Fig. 3.1 is presented in section 3.6.

3.2 Introduction of the secondary clamp circuit

In the Chapter 2, a high-frequency transformer isolated adjustable speed drive was introduced and a new modulation strategy was proposed for it. The switching pulses for this PET are show in Fig. 3.2. When a zero vector is applied in the matrix converter, the output currents free-wheel through the matrix converter and the transformer currents

37



Figure 3.2: Switching pulses

are brought to zero with the help of the secondary side clamp circuit, hence S_1 and S_2 are soft switched (ZCS) and the primary clamp circuit is no longer necessary.

Whenever there is a change in switching state of the matrix converter, the currents through the finite leakage inductances of the transformer are interrupted and the secondary clamp circuit provides a path for these currents to flow. The clamp circuit is operational four times each switching cycle, for two zero vector to active vector transitions (t_{za1}, t_{za2}) and remaining two for active vector to zero vector transitions (t_{az1}, t_{az2}) . In this chapter, the clamp circuit refers to the secondary side clamp circuit. This secondary clamp circuit consists of a diode bridge connected to a capacitor. The power extracted from the clamp circuit can be supplied to the auxiliary control circuits. For simplicity here, the clamp capacitor is replaced by a DC voltage source of magnitude V_{clp} . The diodes in the clamp circuit are considered to be fast acting so they come into conduction almost instantaneously.

3.3 Clamp circuit analysis

In this analysis, the turns ratio of the three winding transformers are assumed to be unity, and the leakage inductance of each winding is L. Also, the commutation time intervals are small compared with the switching period. The input three-phase balanced voltages have a peak value of V_i and a frequency of ω_i rad/sec and are given by (3.1). Neglecting the switching ripple, the output current can be represented by (3.2), where, I_o is the peak value of output current and ω_o is the frequency in rad/sec.

$$\begin{aligned} w_a(t) &= V_i \cos(\omega_i t) \\ v_b(t) &= V_i \cos\left(\omega_i t - \frac{2\pi}{3}\right) \\ v_c(t) &= V_i \cos\left(\omega_i t + \frac{2\pi}{3}\right) \end{aligned}$$
(3.1)

$$i_u(t) = I_o \cos(\omega_o t - \phi)$$

$$i_v(t) = I_o \cos\left(\omega_o t - \frac{2\pi}{3} - \phi\right)$$

$$i_w(t) = I_o \cos\left(\omega_o t + \frac{2\pi}{3} - \phi\right)$$
(3.2)

In this section the operation of the secondary clamp circuit is analyzed for 1) an active to zero vector transition and 2) a zero to active vector transition. Depending on the polarity of output currents, each of these transitions has two special cases, a) when two out of three currents are positive and b) when two out of three-phase currents are negative. Hence, each transition is studied when $\omega_o t$ is between $-\pi/6$ and $\pi/6$ and when $\omega_o t$ is between $\pi/6$ and $\pi/2$ (assuming $\phi = 0$). The analysis for these commutation intervals when S_2 is on, are the same as for the commutation intervals when S_1 is on, except that the input voltage polarities are reversed, hence, the analysis when S_1 is on is sufficient.

As the switching frequency is much higher than the fundamental frequency, in one switching cycle, the output currents and the input phase voltages can be considered to be DC. The output phase currents are assumed to be constant current sources of values I_u , I_v and I_w . As synchronously rotating vector are used in the modulation, each output terminal of the matrix converter is connected to a unique input terminal, leading to six possible combinations. When an active vector is applied, for generality, the input voltage connected to phase u, v and w are denoted by v_u , v_v and v_w respectively. For a given choice of output currents, the power loss and commutation time depends on the input voltage. This dependence can be observed by plotting the clamp power loss or the commutation time required for all combinations of input voltages. In the last part of this section, the voltage loss incurred due to operation of clamp circuit is calculated.



Figure 3.3: Active vector to zero vector equivalent circuit

3.3.1 Active to zero vector transition

When a zero vector is applied in the matrix converter, the output currents free-wheel through the matrix converter and the clamp circuit will provide a path for the leakage currents to flow. Without loss of generality, let us analyze the case when S_1 is on and a zero vector is applied (t_{az1}) . Neglecting the magnetizing current, the transformer can be represented by its series leakage inductance (2L) only. Say i_u is positive and i_v and i_w are negative (i.e $-\pi/6 \leq \omega_o t \leq \pi/6$). The instantaneous values of phase currents i_u , i_v and i_w when the zero vector is applied are denoted by I_u , I_v and I_w respectively. The positive direction of current in phase u forces diode d_{u_1} to conduct. Similarly, diodes d_{v_2} and d_{w_2} will conduct. The equivalent circuit when the clamp circuit is operational is drawn in Fig. 3.3. The KVL equation for the loop containing v_u , 2L, d_{u_1} , V_{clp} , d_{v_2} , 2L and v_w is given by (3.4). The KCL equation for currents at node n_i results in (3.5).

$$v_u - 2L\frac{d}{dt}i_u - V_{clp} + 2L\frac{d}{dt}i_v - v_v = 0$$
(3.3)

$$v_u - 2L\frac{d}{dt}i_u - V_{clp} + 2L\frac{d}{dt}i_w - v_w = 0$$
(3.4)

$$\frac{d}{dt}i_u + \frac{d}{dt}i_v + \frac{d}{dt}i_w = 0 \tag{3.5}$$

The above three equations can be solved to obtain the rate of change of phase currents (3.6) - (3.8)

$$\frac{d}{dt}i_u = \frac{3v_u - 2V_{clp}}{6L} \tag{3.6}$$

$$\frac{d}{dt}i_v = \frac{3v_v + V_{clp}}{6L} \tag{3.7}$$

$$\frac{d}{dt}i_w = \frac{3v_w + V_{clp}}{6L} \tag{3.8}$$

The currents i_u , i_v and i_w are flowing with rates defined by equations (3.6) - (3.8). Depending on the values of t_v and t_w from Table 3.1, either i_v or i_w will become zero amperes. In this case, let current i_v reaches zero amperes first. Diode d_{v_2} will stop conducting and equation (3.3) is no longer involved. Solving for the new rate of change of currents, to obtain (3.9).

$$\frac{d}{dt}i_{u} = -\frac{d}{dt}i_{w} = \frac{v_{u} - v_{w} - V_{clp}}{4L}$$
(3.9)

Once the initial value of current as well as rate of change of currents is known, the total time required (t_{ct}) for all the phase currents to become zero can be calculated. For this case, the current that flows into the clamp circuit is the same as i_u . The power loss in the clamp circuit can be calculated by (3.10), where $\langle i_u \rangle_{t_{ct}}$ is the average value of current i_u during the commutation interval t_{ct} . The clamp circuit power loss for the case when i_w reaches zero before i_v can be found by interchanging I_v with I_w and v_v with v_w in (3.10).

$$P_{Clamp} = V_{clp} < i_u >_{t_{ct}} = \frac{V_{clp} f_s L}{2} \left[-\frac{(I_u - I_w)^2}{v_u - v_w - V_{clp}} + \frac{3I_v^2}{3v_v + V_{clp}} \right]$$
(3.10)

A similar analysis is done when S_1 is on and i_u and i_v are positive and i_w is negative $(\pi/6 \le \omega_o t \le \pi/2)$. The power loss for interval t_{za1} when i_v reaches zero first is given in (3.11).

$$P_{Clamp} = V_{clp} < i_w >_{t_{ct}}$$
$$= \frac{V_{clp} f_s L}{2} \left[-\frac{(I_u - I_w)^2}{v_u - v_w - V_{clp}} + \frac{3I_v^2}{-3v_v + V_{clp}} \right]$$
(3.11)

From the above analysis, it is clear that as long as the clamp capacitor voltage is maintained higher than three times the peak line to neutral voltage, the currents will change in the desired manner. If one of the currents reach zero, the other two currents flow at a different rate and eventually become zero. The power loss in the clamp circuit is proportional to switching frequency, leakage inductance and the instantaneous values of phase currents and voltages. The value of V_{clp} is selected to be $5 \times V_i$. For selected values of output currents, the the per unit power loss is plotted as $\omega_i t$ is varied from





Figure 3.4: Active to zero vector $\omega_o t(\text{rads}) = -\frac{\pi}{6}$ (P), $-\frac{\pi}{12}$ (Q) , 0 (R), $\frac{\pi}{12}$ (S) and $\frac{\pi}{6}$ (T)

Figure 3.5: Active to zero vector $\omega_o t(\text{rads}) = \frac{\pi}{6} \text{ (P)}, \frac{\pi}{4} \text{ (Q)}, \frac{\pi}{3} \text{ (R)}, \frac{5\pi}{12}$ (S) and $\frac{\pi}{2}$ (T)

		(, i)
Output currents	time	Total time (t_{ct})
$-\frac{\pi}{6} \le \omega_o t \le \frac{\pi}{6}$	$t_v = -\frac{6LI_v}{3v_v + V_{clp}}$	$= -2L \frac{(I_u - I_w)}{v_u - v_w - V_{clp}} \text{for } t_v \le t_w$
	$t_w = -\frac{6LI_w}{3v_w + V_{clp}}$	$= -2L \frac{(I_u - I_v)}{v_u - v_v - V_{clp}}$ otherwise
$\frac{\pi}{6} \le \omega_o t \le \frac{\pi}{2}$	$t_v = -\frac{6LI_v}{3v_v - V_{clp}}$	$= -2L \frac{(I_u - I_w)}{v_u - v_w - V_{clp}} \text{for } t_v \le t_u$
	$t_u = -\frac{6LI_u}{3v_u - V_{clp}}$	$= -2L \frac{(I_v - I_w)}{v_v - v_w - V_{clp}}$ otherwise

Table 3.1: Commutation time (active to zero, S_1 on)

0 to 2π in Fig. 3.4 and Fig. 3.5. One per unit value is equal to $0.5LI_o^2 f_s$ Watts. It is observed that the power loss for the case when $-\pi/6 \leq \omega_o t \leq \pi/6$ is bound between the curves corresponding to $\omega_o t = -\pi/6$ and $\omega_o t = \pi/6$. When $\pi/6 \leq \omega_o t \leq \pi/2$, the clamp power loss is bounded by the power loss curves corresponding to $\omega_o t = \pi/6$ and $\omega_o t = \pi/2$. From the graph, the peak value of power that goes into the clamp circuit is 4.6 per unit and the minimum value is 2.2 per unit.

Table 3.1 lists the commutation time for an active to zero vector transition for different cases of output currents. The per unit time required is plotted in Fig. 3.6 for five selected values for output current for different combinations of input voltages. The maximum commutation required is equal to 1.06 p.u., where one p.u. equals LI_o/V_i .



Figure 3.6: Commutation time required: Active to zero vector $\omega_o t$ (rads) = $-\frac{\pi}{6}$ (P), 0 (Q), $\frac{\pi}{6}$ (R), $\frac{\pi}{3}$ (S) and $\frac{\pi}{2}$ (T)

3.3.2 Zero vector to active vector transition

A similar analysis is done for the transition from a zero vector to an active vector when switch S_1 is on (t_{za1}) . Consider the currents in the transformer windings are zero, when an active vector is applied in the matrix converter such that phase u, v and ware connected to phase voltages v_u, v_v and v_w respectively. In one switching cycle, the output currents can be assumed to be constant and are given by I_u , I_v and I_w for phase u, v and w respectively. The clamp circuit is active until the transformer currents match the value of the output currents. For the case when current I_u is positive and I_v and I_w are negative ($\pi/6 \leq \omega_o t \leq \pi/6$), diode d_{u_2}, d_{v_1} and d_{w_1} will begin to conduct. The equivalent circuit is drawn in Fig. 3.7. The KVL equations for the two loops in this circuit are given by (3.12) and (3.13). The KCL equation at node n_i is given by (3.14).

$$v_v - 2L\frac{d}{dt}i_v - V_{clp} + 2L\frac{d}{dt}i_u - v_u = 0$$
(3.12)

$$v_w - 2L\frac{d}{dt}i_w - V_{clp} + 2L\frac{d}{dt}i_u - v_u = 0$$
(3.13)

$$\frac{d}{dt}i_u + \frac{d}{dt}i_v + \frac{d}{dt}i_w = 0$$
(3.14)



Figure 3.7: Zero vector to active vector equivalent circuit

The three equations and three unknowns (3.12) - (3.14) can be solved to obtain the rate of change of currents given by (3.15) - (3.17).

$$\frac{d}{dt}i_u = \frac{3v_u + 2V_{clp}}{6L} \tag{3.15}$$

$$\frac{d}{dt}i_v = \frac{3v_v - V_{clp}}{6L} \tag{3.16}$$

$$\frac{d}{dt}i_w = \frac{3v_w - V_{clp}}{6L} \tag{3.17}$$

The phase currents i_u , i_v and i_w currents flow at the rates defined above until the time one of them reaches the output value. If i_v equals I_v , then, diode d_{v_1} ceases to conduct and i_u and i_w will change at rates given by (3.18).

$$\frac{d}{dt}i_{u} = -\frac{d}{dt}i_{w} = \frac{v_{u} - v_{w} + V_{clp}}{4L}$$
(3.18)

During the commutation interval, the difference between the inductor current and output current flows into the the clamp circuit. The associated clamp power loss is given by (3.19). When I_u and I_v are positive and I_w is negative $(\pi/6 \le \omega_o t \le \pi/2)$, the clamp power loss is given by (3.20).

$$P_{Clamp} = V_{clp}(I_u - \langle i_u \rangle_{t_{ct}})$$

= $\frac{V_{clp}f_s L}{2} \left[\frac{(I_u - I_w)^2}{v_u - v_w + V_{clp}} - \frac{3I_v^2}{3v_v - V_{clp}} \right]$ (3.19)

44





Figure 3.8: Zero to active vector $\omega_o t$ (rads)= $-\frac{\pi}{6}$ (P), $-\frac{\pi}{12}$ (Q), 0 (R), $\frac{\pi}{12}$ (S) and $\frac{\pi}{6}$ (T)

Figure 3.9: Zero to active vector $\omega_o t$ (rads) = $\frac{\pi}{6}$ (P) , $\frac{\pi}{4}$ (Q), $\frac{\pi}{3}$ (R), $\frac{5\pi}{12}$ (S) and $\frac{\pi}{2}$ (T)

Table 3.2: Commutation time for zero to active vector with S_1 on)

Transition	time	Total time (t_{ct})
$\pi < \dots + < \pi$	$t_v = \frac{6LI_v}{3v_v - V_{clp}}$	$= 2L \frac{(I_u - I_w)}{v_u - v_w + V_{clp}} \text{ for } t_v \le t_w$
$-6 \leq \omega_0 \iota \leq 6$	$t_w = \frac{6LI_w}{3v_w - V_{clp}}$	$= 2L \frac{(I_u - I_v)}{v_u - v_v + V_{clp}}$ otherwise
$\frac{\pi}{2} < 0, t < \frac{\pi}{2}$	$t_v = \frac{6LI_v}{3v_v + V_{clp}}$	$= 2L \frac{(I_u - I_w)}{v_u - v_w + V_{clp}} \text{for } t_v \le t_u$
$\overline{6} \stackrel{\scriptstyle \sim}{=} \omega_0 \iota \stackrel{\scriptstyle \sim}{=} \overline{2}$	$t_u = \frac{6LI_u}{3v_u + V_{clp}}$	$=2L\frac{(I_v-I_w)}{v_v-v_w+V_{clp}}$ otherwise

$$P_{Clamp} = V_{clp}(I_w - \langle i_w \rangle_{t_{ct}})$$

= $\frac{V_{clp} f_s L}{2} \left[\frac{(I_u - I_w)^2}{v_u - v_w + V_{clp}} + \frac{3I_v^2}{3v_v + V_{clp}} \right]$ (3.20)

The value of V_{clp} is selected to be $5 \times V_i$. The per unit power loss curves for certain values of output current as $\omega_o t$ is varied from 0 to 2π are plotted in Fig. 3.8 and Fig. 3.9. Here again the maximum and minimum clamp power loss is equal to 4.6 and 2.2 p.u. respectively. Table 3.2 lists the commutation time required for a zero to active vector transition. As before, the commutation time required in per unit as a function of input voltage is plotted in Fig. 3.10. The worst case commutation time required is $1.06LI_o/V_i$ sec. There is some output voltage distortion associated with the clamp



Figure 3.10: Commutation time required: Zero to active vector $\omega_o t$ (rads) = $-\frac{\pi}{6}$ (P), 0 (Q), $\frac{\pi}{6}$ (R), $\frac{\pi}{3}$ (S) and $\frac{\pi}{2}$ (T)

circuit operation. During an active to zero vector transition, the clamp circuit voltage does not appear across the load. A voltage loss or distortion occurs only during a zero to active vector transitions. In a worst case situation, the clamp circuit may apply $\pm \frac{2}{3}V_{clp}$ across the output line-neutral of a particular phase for the longest commutation time.

3.4 Simulation results

3.4.1 Intervals t_{az1} and t_{za1}

The circuit in Fig. 3.1 is simulated in SABER® for one switching cycle. The input voltages and output currents are DC sources of magnitudes listed in Table 3.3. Fig. 3.11 is the plot of currents i_u , i_v and i_w when a zero vector is applied and S_1 is on. The phase currents for the zero to active vector transition when S_1 is on are shown in Fig. 3.12. The simulated values for the slope of the current coincide with the calculated values. The clamp circuit power loss during a zero to active vector transition when S_1 is on (t_{za2}) is equal to the power loss for an active to zero transition when S_1 is on (t_{az1}) and it is equal to 44 Watt by calculation. The clamp circuit power loss for the transition t_{az2} equals that of t_{za1} equal to 25.74 Watts. The values for commutation time required as well as total clamp power loss in one switching cycle are listed in 3.4. The calculated values match the simulated values very closely.

Table 3.3: Simulation parameters

	100 40 CO V
v_u, v_v, v_w	100 ,-40,-60 V
I_u, I_v, I_w	20, -5,-15 A
V_{clp}	600 V
L	50μ H
Turns ratio	1:1:1



Figure 3.11: Active to zero vector simulation results (t_{az1})



Figure 3.12: Zero to active vector simulation results (t_{za1})

Value	Simulation	Calculation
P_{clamp}	$135.5 \mathrm{~W}$	$139.4~\mathrm{W}$
$t_{az1} = t_{za2}$	8.05μ sec	$7.95 \mu \mathrm{sec}$
$t_{za1} = t_{az2}$	4.62μ sec	4.61μ sec

Table 3.4: Simulation results



Table 3.5: Simulation parameters: Three-phase system

Figure 3.13: Three-phase AC-AC PET(1kW, 0.8 pf)

24m

time (sec)

28m

32m

3.4.2 Three-phase AC-AC case for same frequency

16m

20m

A 1kW three-phase AC-AC system in Fig. 3.1 is simulated with the parameters listed in Table 3.5. Where V_i and V_o are the peak values of input and output balanced three-phase voltages respectively. The matrix converter is controlled using the technique described in section 2.2. Using this method, unity power factor is obtained on the primary side. The input voltage and filtered input current for phase a and current for phase u are plotted in Fig. 3.13. The worst case clamp power loss according to the analysis is 81.154 W (8.12 %) and the least clamp power loss is 17.68 W (1.768 %). The clamp power loss from simulation equals 79.89 W (7.99 %). The peak output voltage from simulation is 26.22 Watts, that corresponds to a voltage loss of 3.78 V. This is below the worst prediction of 10.27 V for voltage loss.

3.4.3 Three-phase variable frequency AC

The same 1kW system is simulated for a reference output voltage of 30V, 120 Hz. The output current and filtered input current along with input voltage for one phase are plotted in Fig. 3.14. The simulated values of power loss and voltage loss are within the calculated limits.



Figure 3.14: 1kW, 0.8 pf, three-phase variable frequency 60Hz - 120Hz

3.5 Experimental results

The experimental setup shown in Fig. 3.15 can be used to test the active vector to zero vector transition as well as the zero vector to active vector transition. The matrix converter is switched between vectors (ABC) and (AAA). Pulses p_1 and p_2 in Fig. 3.15 are generated in the FPGA and have a frequency of $f_s = 1000Hz$ and 50% duty. A dead-time (t_{dt}) of 1.5μ sec is introduced between the pulses p_1 and p_2 . Switches S_{Av}, S_{Aw} are provided pulses p_1 and switches S_{Bv}, S_{Cw} are provided pulses p_2 . S_{Au} is provided pulses $p_1||p_2$. The input voltages are supplied by two regulated DC supplies connected in series. By changing the DC voltages, two distinct sectors can be observed, one when two out of three-phase currents are positive and the other when only one phase current is positive. The f are in Table. 3.6. The inductance is made larger than in a high-frequency transformer so as to better observe the transitions. The clamp voltage V_{clp1} is set to 80V using a regulated DC supply.



Figure 3.15: Experimental setup for active to zero vector transition

V_{AB}, V_{BC}	33V, 10V(Sector 1); 10V, 33V(Sector 2)
f_s	1kHz
RL Load	$6.6\Omega, 51.3 \mathrm{mH}$
2L	$500 \mu H$
V_{clp1}	80V

Table 3.6: Experimental setup: Parameters

The currents through the input inductance, i_A , i_B and i_C are measured and compared with simulation in PLECS®. The simulation and experimental results in Sector 1 are in Fig. 3.16. The simulation and experimental results when the current is in Sector 2 are in Fig. 3.17. These results for a zero vector to active vector transition for Sector 1 and 2 are summarized in Table. 3.7. The simulation and hardware results for an active vector to zero vector transition are in Table. 3.8.

It can be seen that the transition times and the currents from the experimental setup and the simulation concur. A dead-time is implemented when switching from one vector to another, hence the load side clamp circuit, V_{clp2} is active every switching transition to conduct the load currents.

3.6 Comparison of dead-time commutation and four-step commutation

In this chapter, the commutation of transformer leakage inductance currents has been discussed. For an active vector to zero vector transition, this analysis assumes that a



Figure 3.16: Zero to active vector transition in Sector 1 (a) hardware (0.5A/div) (c) simulation; Active to zero vector transition in Sector 1 (b) hardware(0.5A/div) (d) simulation.

	Sector 1		Sector 2			
Parameter	Hardware	Simulation	Hardware	Simulation		
i_A (A)	1.5	1.51	1.24	1.08		
i_B (A)	-0.4	-0.43	0.39	0.43		
i_C (A)	-1.1	-1.08	-1.57	-1.51		
$t_1 \ (\mu s)$	6.66	6.07	4.97	6.07		
$t_1 + t_2 \ (\mu s)$	10.41	10.36	10.14	10.36		

Table 3.7: Experimental results: Zero to active vector transition



Figure 3.17: Zero to active vector transition in Sector 2 (a) hardware (0.5A/div) (c) simulation; Active to zero vector transition in Sector 2 (b) hardware(0.5A/div) (d) simulation.

	Sector 1		Sector 2		
Parameter	Hardware	Simulation	Hardware	Simulation	
i_A (A)	1.5	1.65	1.07	1.17	
i_B (A)	-0.42	-0.47	0.36	0.47	
i_C (A)	-1.235	-1.18	-1.59	-1.65	
$t_1 \ (\mu s)$	11.439	11.91	9.63	11.91	
$t_1 + t_2 \; (\mu s)$	34.85	36.18	32.27	36.18	

Table 3.8: Experimental results: Active to zero vector transition

zero vector has been applied in the matrix converter. For a zero vector to active vector transition, it assumes that the active vector has been applied in the converter. However, power electronic devices have finite turn-on and turn-off times, hence commutation of switches in the matrix converter needs to be considered.

In Matrix converters, commutation of switches is done either using dead-time or fourstep commutation. Four-step commutation is a well known method for commutation of four-quadrant switches in a matrix converter when the input is a voltage port and the output a current port [59]. In [15] four-step commutation is implemented in a PET that satisfies this condition. The input as well as output ports of the MC in Fig. 3.1 are inductive in nature. A comparison of dead-time commutation and four-step commutation for the MC based PET in Fig. 3.1 for Carrier Based Modulation as well as Space Vector Modulation is presented in this section.

3.6.1 Simulation results

In order to analyze the benefits of four-step commutation, the circuit in Fig. 3.18 is simulated in PLECS(\mathbb{R}) for three transitions: active vector (ABC) to active vector (BCA), active vector (ABC) to zero vector (AAA) and zero vector (AAA) to active vector (ABC) transition. The input phase voltages V_{AB} and V_{BC} are set to 30V and 40V respectively. The inductors $2L = 15\mu H$ represent the equivalent leakage inductance of the high-frequency transformer. The output currents are set to 10A, 5A and -15A for phase u, v, and w respectively. The four steps of the commutation process occur at time 4, 6, 8 and 10 μ s. For dead-time commutation, t_{dt} lasts from 4 μ s to 10 μ s. Consider the transition from vector (ABC) to (BCA) as shown in Fig. 3.19(a) and 3.19(b). The blue, green and red waveforms are for phase a, b and c respectively. If dead-time commutation is implemented, the input side clamp (i_{a1}, i_{b1}, i_{c1}) circuit is operational for the time required to commutate the currents to zero at the start of the dead-time. As shown in Fig. 3.19(a), the output side clamp circuit (i_{a2}, i_{b2}, i_{c2}) is operational for the entire dead-time t_{dt} in addition to the time required to commutate the currents once (BCA) is applied in the converter. If four-step commutation is implemented, when the correct switch in the incoming branch is turned on, the phase currents commutate naturally if the voltage polarity is correct. When the outgoing branch is completely turned off at 8μ s, the remaining currents commutate and reach their desired values. Clamp currents



Figure 3.18: Simulation circuit for a transition from (ABC) to (BCA)

flow for an interval (t_{com}) to help with the commutation process as shown in Fig. 3.19(b). The value of t_{com} is a function of 2L, V_{clp} and input voltages. The simulation results for an active vector (ABC) to zero vector (AAA) transition as well as for a zero vector (AAA) to active vector (ABC) transition for dead-time and four-step commutation are given in Fig. 3.19(c), 3.19(e) and 3.19(d), 3.19(f) respectively.

During the dead-time, all the load current flows through the output clamp circuit. Hence the clamp voltage, V_{clp} appears across the load voltage terminals. This leads to a reduction in the output voltages seen by the load as well as voltage distortion. The maximum clamp voltage seen between any two phases is $\pm V_{clp}$. This voltage appears 4 times every switching cycle when SVM is used and 6 times when CBM is used. Neglecting the voltage loss associated with t_{com} , the maximum loss in line to line output voltage due to a dead-time of t_{dt} for a switching frequency of f_s is $4t_{dt}f_sV_{clp}$ V per phase when SVM is used and $6t_{dt}f_sV_{clp}$ V when carrier-based modulation is used (worst case with 6 vectors transitions each cycle). It is concluded that dead-time commutation will lead to excessive loss in output voltage, voltage distortion, as well as power loss.

3.6.2 Experimental results

The parameters for the experimental results are listed in Table 3.9. The experimental results when CCW and CW vectors are used for equal duration when CBM is employed with dead-time and four-step commutation are in Fig. 3.20(a) and 3.20(b) respectively. The experimental results when SVM is employed with dead-time commutation and four-step commutation are in Fig. 3.20(c) and 3.20(d) respectively.



Figure 3.19: Comparison of dead-time and four-step commutation: with (a), (c), (e) dead-time commutation; (b), (d), (f) four-step commutation.


(c) SVM:Dead-time commutation



Figure 3.20: Experimental results using CCW and CW vectors for equal duration: Carrier based modulation with (a) dead-time and (b) four-step commutation; Space Vector Modulation with (c) dead-time and (d) four-step commutation. (top) Input voltage v_{an_i} (C3: 50V/div) and current i_{af} (C1:1A/div); (bottom) output voltage v_{un_o} (C4:50V/div) and current i_u (C2:1A/div), time (5ms/div).

Table 3.9: Experimental parameters: Comparison of commutation techniques

V_i, V_o	$40\sqrt{2}V$, 22.23V
$\omega_i \ , \omega_o \ , f_s$	$2\pi 60 \text{ rad/s}, 2\pi 50 \text{ rad/s}, 10 \text{kHz}$
R, L , L_f , C_f	5.5Ω , 25mH , 0.5mH, 20 $\mu F(star)$

When the load is not connected, the fundamental component of output line-neutral voltage is 16.03V. When the PET is loaded, the observed line-neutral voltages are listed in Table. 3.10. When dead-time commutation is employed the worst case voltage loss expected when CBM is used is 11.4V for a clamp voltage of around 190V. The observed voltage loss is less than that and equals 8.4V. When SVM is employed, the worst case expected voltage loss is 7.6V and the observed loss is 4.5V. The maximum voltage is observed when SVM is applied with four-step commutation. As four-step commutation is suspended when the current direction is not accurately known, there is some voltage loss of 1.65V associated with it. At higher currents, four-step commutation is possible for a higher percentage of time, leading to better voltage profiles. When four-step commutation is used with carrier based modulation, there is a voltage loss of 3.67V. This is because there are six switching transitions as compared to four switching transitions per cycle when SVM is used.

The Fourier spectrum of the output current for phase u for SVM is in Fig. 3.21(a). It is observed that the component of output current at the fundamental frequency is higher for four-step commutation (1.38A) as compared to dead-time commutation (1.12A) when SVM is used. The Total Waveform Distortion (TWD) is given by (3.21). The TWD for I_u when dead-time commutation is used is 5.61% and when four-step commutation is used is 4.4%. The Fourier spectrum of the output voltage, v_{un_o} is in Fig. 3.21(b). It is observed that the fundamental component of the output voltage is higher when four-step commutation is used. The output voltage contains harmonics at the fundamental frequency, at 5kHz (because CCW and CW vectors are used) as well as harmonics at the switching frequency of 10kHz. In conclusion, four-step commutation leads to better output TWD, lesser power loss in the clamp circuit and consequently better efficiency.

$$TWD = \frac{\sqrt{I_{rms}^2 - I_{f1}^2}}{I_{f1}} \tag{3.21}$$

	CBM	CBM	SVM	SVM	
	dead-time	four-step	dead-time	four-step	
V_{un_o} RMS (V)	7.60	12.36	11.53	14.38	
I_u RMS (A)	0.74	1.19	1.12	1.38	
TWD i_u %	6.75	4.83	5.61	4.40	

Table 3.10: Comparison of modulation and commutation



Figure 3.21: Fourier spectrum of output current and voltage

3.7 Conclusions

In this chapter, the secondary side clamp circuit requirement is studied and a detailed analysis of its operation is done. The lower and upper limit of power loss associated with the leakage energy commutation using a clamp circuit is calculated and these values concur with the simulated values. The following conclusions can be drawn,

- 1. The secondary clamp voltage must be greater than three times the peak line to neutral voltage.
- 2. The clamp circuit is operational every time the load side converter is switched.
- 3. For $V_{clp} = 5 \times V_i$, the power loss associated with it is between $4.46 f_s L I_o^2$ and

 $9.18 f_s LI_o^2$ Watts.

- 4. For $V_{clp} = 5 \times V_i$, the maximum time required for commutation can be upto $1.06 \times LI_o/V_i$ sec.
- 5. Zero to active vector transitions and dead-time in the load side converter lead to distortion in the output voltage waveform.
- 6. This research proves that even though the MC has leakage inductance on the input port, four-step commutation instead of dead-time commutation leads to better voltage profile and consequently better output currents. It also leads to improved efficiency of the PET.

The results of this analysis can be used to design the resistive and capacitive components for the clamp circuit. This analysis is very general and can be further extended to the clamp circuit operation in any other relevant topologies.

Note: Parts of this chapter have been reprinted from [36] ©2010 IEEE and [60]

Chapter 4

Single-Phase AC-DC Power Electronic Transformer

The switching frequency in high-frequency transformer (HFT) isolated AD-DC converters is limited by the device stress, switching losses and commutation time for leakage energy. Hence, converters having soft-switching characteristics are being explored.

Dual Active Bridge Converters (DABC) are a class of high-frequency transformerisolated converters that have inherent qualities of soft switching, low device stress, small filter requirements and use of the HFT inductance for power transfer. These make them suitable for applications requiring high power density. DABC was introduced for DC-DC power conversion in [51,52]; In this converter, two square-wave voltages are phase shift modulated (PSM) to control the power flow across the leakage inductance of the transformer. When the input and output voltages are not the same value, the soft switching range of PSM DABC is limited. New modulations methods, where one or both the converters are duty-ratio modulated (PWM) were proposed to extend this soft switching range [61]. These PWM techniques can be categorized into inner mode, outer mode, Trapezoidal and Triangular current modulation [62–67]. Triangular current mode modulation is used when the input and output voltages are not equal; this strategy results in maximum ZVS/ZCS at the cost of high RMS currents [68]. In trapezoidal current modulation, low RMS currents are obtained in the transformer and higher power transfer is possible compared to triangular current modulation. Triangular current



Figure 4.1: Single-phase AC-DC PET

mode modulation is a special case of the 'inner mode' of operation. The 'inner mode' of operation is characterized by limiting the PWM pulses of the high voltage side converter to be contained within those of the low voltage side converter. It is possible to obtain ZCS in the low voltage side converter in this mode of operation. Single-phase AC-DC converters based on DAB principle have been proposed in [69–72].

In this chapter, a control method for a single-phase AC-DC converter (PET) (shown in Fig. 4.1) that is based on the 'inner mode' DAB principle is proposed that simultaneously has the following features: a) galvanic isolation, b) bi-directional power flow, c) ZCS for the primary side switches and ZVS turn-on for the secondary side switches, d) linear power relationship for easy control implementation, e) unity power factor under open-loop control and f) single-stage power conversion. In order to analyze the singlephase AC-DC converter, first, the 'inner mode' of operation for a DC-DC DABC with the primary switches in push-pull topology is analyzed in section 4.1.1. This analysis is extended to single-phase AC-DC converter in section 4.1.2. In section 4.2, the simulation results in SABER®are compared with the theoretical analysis. The experimental results for the DC-DC converter and single-phase AC-DC converter are compared with the simulation results in section 4.4 and 4.5 respectively.

4.1 Topology and modulation

An AC-DC Dual Active Bridge Converter (DABC) with the primary side in pushpull configuration is shown in Fig. 4.1. An AC voltage source $v_i = \hat{V}_i \sin(\omega t)$, where $\omega = 2\pi f$ is connected to the primary of a three-winding high-frequency transformer and a DC voltage source V_o is connected to the secondary of the high-frequency transformer through a H-bridge inverter. The turns-ratio of one half of the primary winding to the secondary winding of the transformer is 1: n. In the following analysis, it is assumed that $V_o \ge n\hat{V}_i$. The leakage inductances of the two primary windings are L_{p_1} and L_{p_2} and that of the secondary winding is L_s . The primary side switches S_1 and S_2 are four-quadrant. They are switched at 50% duty-ratio in a complementary way at a switching frequency of $f_s(=\frac{1}{T_s})$. If the leakage inductance of the two primary windings are identical, $L_{p_1} = L_{p_2} = L_p$, and the magnetizing current is neglected, the equivalent circuit of the converter system seen from the secondary side of the transformer is shown in Fig. 4.2 where, $L = n^2 L_p + L_s$. The voltage v_p is $+nv_i$ when S_1 is on and is $-nv_i$ when S_2 is on. In one switching cycle, the net volt-sec applied to the transformer is zero; thus, the flux in the transformer core is balanced in the modulation technique described below. The secondary side switches S'_{o_1} and S'_{o_2} are switched complementary to S_{o_1} and S_{o_2} respectively. The voltage v_s can be controlled to be $+V_o$, $-V_o$ or 0.

4.1.1 Analysis of DC-DC converter

In a DC-DC DABC, the input voltage in Fig. 4.1 is constant at V_i . The switches S_1 and S_2 can be two-quadrant. The secondary side converter is pulse-width modulated with a duty-ratio given by (4.1). A phase shift of $\delta \frac{T_s}{2}$ is introduced between the voltages v_p and v_s to obtain power transfer (Fig.4.3 (a)). In the 'inner mode' of operation, the absolute value of δ is limited by (4.2) [67]. This modulation technique ensures that the average voltage applied across the inductor is zero every $T_s/2$ and the switches S_1 and S_2 are switched at zero current, thus reducing the switching losses considerably. An added advantage being a snubber circuit is no longer necessary for these switches. The inductor current i_L , i_o and i_i are shown in Fig. 4.3 (c), (d) and (e) respectively.

$$d = \frac{nV_i}{V_o} \tag{4.1}$$

$$0 \le |\delta| \le (1-d)\frac{1}{2} \tag{4.2}$$

Each half cycle $(T_s/2)$ is divided into three time intervals, t_1 , t_2 and t_3 given by (4.3). The current in the inductor during these time intervals is given by, (4.4),(4.5) and (4.6). Solving equations (4.1)- (4.6), the values of I_0 and I_3 equal zero (eq. (4.7)).



Figure 4.2: Equivalent circuit of single-phase AC-DC converter

$$t_{1} = \frac{T_{s}}{4} + \delta \frac{T_{s}}{2} - d \frac{T_{s}}{4}$$

$$t_{2} = d \frac{T_{s}}{2}$$

$$t_{3} = \frac{T_{s}}{2} - t_{1} - t_{2}$$
(4.3)

$$I_1 = I_0 + \frac{nV_i}{L}t_1$$
 (4.4)

$$I_2 = I_1 + \frac{(nV_i - V_o)}{L} t_2 \tag{4.5}$$

$$I_3 = I_2 + \frac{nV_i}{L} t_3 \tag{4.6}$$

$$I_0 = I_3 = 0 (4.7)$$

The secondary side converter can be modulated by two switching strategies; in the first case (Fig. 4.3 (a)), the zero voltage is applied by turning on S'_{o_1} and S'_{o_2} . In the second method (Fig. 4.3 (b)), the zero voltage is applied either by S_{o_1} and S_{o_2} or by S'_{o_1} and S'_{o_2} ; this method has an advantage that the secondary side pulses are square wave. A dead-time is introduced between the transition of switches in the same leg. The switches in the secondary side turn on under ZVS condition as anti-parallel diodes are conducting when these switches are turned on. This condition is true for any direction of power flow. The voltage, current and power bases are selected as (4.8). The average power supplied is calculated to be (4.10). A maximum power of 0.233 pu can be transferred using this converter when d = 2/3. Due to the constraints on phase shift δ , the power transferred in the inner mode of operation is lesser than a PSM



Figure 4.3: Switching waveforms for DC-DC converter

DABC [51](0.523 pu at d=2/3).

$$P_{o} = \left(V_{o}\frac{2}{T_{s}}\right)\left(\frac{I_{1}+I_{2}}{2}\right)t_{2}$$

$$P_{io} = \delta d\frac{nV_{i}V_{o}}{2Lf_{s}} = \delta d^{2}\frac{V_{o}^{2}}{2Lf_{s}}$$

$$(4.9)$$

$$=\delta d^2\pi$$
 ... in pu (4.10)

The per-unit (pu) RMS currents in the transformer are calculated to be (4.11) and (4.12). The primary and secondary RMS voltages are calculated in (4.13) and (4.14) respectively. Using these values the transformer per-unit VA can be calculated as (4.15). The per-unit values of P_t are plotted as a function of P_{io} in Fig. 4.4. A transformer per-unit VA of $P_t = 0.299$ pu is required to transfer the maximum power of 0.233 pu, which corresponds to a transformer utilization (P_{io}/P_t) of 0.779, This is higher than the transformer utilization (0.636) for maximum power transfer at the diode bridge boundary for topology B in [51].

$$\overline{I}_{L_s} = K\sqrt{(1 - 2d + d^2 + 12\delta^2)d^2}$$
(4.11)

where, $K = \frac{2\pi}{4\sqrt{3}}$

$$\bar{I}_{L_{p_1}} = \bar{I}_{L_{p_2}} = \frac{\bar{I}_{L_s}}{\sqrt{2}}$$
(4.12)

$$\overline{V}_p = \frac{d}{\sqrt{2}} \tag{4.13}$$

$$\overline{V}_s = \sqrt{d} \tag{4.14}$$

$$P_t = \frac{1}{2} [2\overline{V}_p \overline{I}_{L_p} + \overline{V}_s \overline{I}_{L_s}]$$

= $\frac{1}{2} (d + \sqrt{d}) \overline{I}_{L_s}$ (4.15)

The size of the DC-side filter capacitor is a function of the RMS ripple current through it. The per-unit value of the RMS ripple current through the DC capacitor, \overline{I}_{rpl} is given





Figure 4.4: DC-DC Converter: P_t Vs P_{io}

Figure 4.5: DC-DC Converter: \overline{I}_{rpl} Vs δ

by (4.17). In Fig. 4.5, the worst case ripple equals 0.256 pu at δ of 0.23 (d=0.54).

$$\overline{I}_o = K\sqrt{(1 - 2d + d^2 + 12\delta^2)d^3}$$
(4.16)

$$\overline{I}_{rpl} = K\sqrt{(1 - 2d + d^2 + 12\delta^2(1 - d))d^3}$$
(4.17)

4.1.2 Analysis of single-phase AC-DC converter

Consider a single-phase phase AC-DC converter in Fig. 4.1. As $f_s \gg f$, in one switching cycle, the input voltage can be approximated as a DC voltage. In the positive half cycle, the ZCS conditions for the primary side switches are satisfied when the output side converter is modulated with a duty ratio of (4.18). In the negative half cycle, the pulses for S_{o_1} and S_{o_2} are interchanged. The duty ratio d of the secondary side converter varies with time, hence the maximum value of δ also varies with time (4.19). Selecting a constant value of δ ensures unity power factor on the AC side. For a given value of $m, |\delta|$ is constrained by (4.20).

$$d(t) = m|\sin(\omega t)| \qquad \cdots \qquad m = \frac{nV_i}{V_o}$$
(4.18)

$$0 \le |\delta| \le \frac{1}{2}(1 - d(t)) \tag{4.19}$$

$$0 \le |\delta| \le \frac{1}{2}(1-m) \tag{4.20}$$

The instantaneous power transferred (4.21) is obtained by replacing V_i and d in (4.10) by $\hat{V}_i \sin(\omega t)$ and (4.18) respectively. From the definition of input power, the switching-cycle averaged input current ($\tilde{i}_i(t)$) can be calculated as (4.22). Under open-loop control, the switching-cycle averaged input current is in phase with the input voltage and unity power factor is obtained. The average power, P_{avg} is calculated by integrating the instantaneous power over one fundamental cycle of the input voltage. The following inferences are made about P_{avg} : 1) The power flow has a linear relationship to δ this simplifies the control strategy where power flow can be bi-directional by changing the sign of δ . 2) As m increases, the range of δ reduces. 3) No power can be transferred when $V_o = n\hat{V}_i$. 4) The locus of maximum power that can be transferred is plotted by the dotted curve in Fig. 4.6. The maximum value of P_{avg} (0.116 pu) occurs when m = 2/3.

$$P(t) = \frac{\delta n^2 \hat{V}_i^2}{2L f_s} \sin^2(\omega t) \tag{4.21}$$

$$\tilde{i}_i(t) = \frac{P(t)}{V_i(t)} = \frac{\delta n^2 \hat{V}_i}{2Lf_s} \sin(\omega t)$$
(4.22)

$$P_{avg} = \frac{1}{2\pi} \int_0^{2\pi} P(\omega t) d(\omega t)$$
(4.23)

$$P_{avg} = \delta n^2 \frac{\hat{V}_i^2}{4Lf_s} = \delta m^2 \frac{V_o^2}{4Lf_s}$$
$$= \frac{1}{2} \delta m^2 \pi \qquad \cdots \text{ in pu}$$
(4.24)



Figure 4.6: AC-DC Converter: P_{avg} Vs δ

$$\overline{I}_{L_s} = K'm\sqrt{9m^2\pi - 64m + 12\pi(1+12\delta^2)}$$
(4.25)

where, K' = 0.1044

$$\overline{I}_{L_{p_1}} = \overline{I}_{L_{p_2}} = \frac{\overline{I}_{L_s}}{\sqrt{2}} \tag{4.26}$$

$$\overline{V}_p = \frac{m}{2} \tag{4.27}$$

$$\overline{V}_s = \sqrt{\frac{2m}{\pi}} \tag{4.28}$$

$$P_t = \frac{1}{2} [2\overline{V}_p \overline{I}_{L_p} + \overline{V}_s \overline{I}_{L_s}]$$
$$= \frac{1}{2} \left(\frac{m}{\sqrt{2}} + \sqrt{\frac{2m}{\pi}} \right) \overline{I}_{L_s}$$
(4.29)

The RMS currents in the transformer windings over one switching period are a function of d(t) and are given by (4.11) and (4.12). The equation for the RMS currents through the transformer windings over one cycle of the input voltage (given by (4.25) and (4.26)) can be found by integrating the the square of (4.11) and (4.12) and determining root of the mean. The RMS values for the primary and secondary voltages are given by (4.27) and (4.28) respectively. The transformer per-unit VA rating is calculated by





Figure 4.7: AC-DC Converter: P_t Vs P_{avg}

Figure 4.8: AC-DC Converter: \overline{I}_{rpl} Vs δ

(4.29). The per-unit values of P_t are plotted as a function of P_{avg} in Fig. 4.7. The dotted curve marks the locus of maximum VA for the transformer for different values of P_{avg} . To transfer the maximum power (0.116 pu), the transformer VA required is 0.176, resulting in a transformer utilization of 0.663. This is the maximum utilization of the transformer. The RMS values for i_o and ripple current in the dc-capacitor given by (4.30) and (4.31) are calculated from (4.16) and (4.17) respectively. The per-unit values of the dc-capacitor current ripple is plotted for selected values of m as a function of δ in Fig. 4.8. The maximum value for \overline{I}_{rpl} (0.185 pu) occurs when δ is 0.21(m=0.58) at a power transfer of 0.111 pu.

$$\overline{I}_o = K'' \sqrt{m^3 (64m^2 - 45m\pi + 80(1+12\delta^2))}$$
(4.30)

$$\overline{I}_{rpl} = K'' \sqrt{m^3 (64m^2 - 45m\pi - 270\pi\delta^2 m + 80 + 960\delta^2)}$$
(4.31)

where, K'' = 0.0661

4.2 Simulation results

The DC-DC converter in Fig. 4.1 is simulated in SABER®. The converter parameters are listed in Table 4.1. The waveforms for the current through S_1 and S_2 as well

Table 4.1: Simulation parameters

	-
$V_i = \hat{V}_i, V_o$	$40\mathrm{V},200\mathrm{V}$
$f_s \;, f$	$5 \rm kHz$, $60 \rm Hz$
$L_{p_1} = L_{p_2} = L_s$	$50 \mu H$
turns-ratio	1:1:1
δ	0.1



Figure 4.9: Simulation result: DC-DC converter

as the input and output currents are shown in Fig. 4.9. As these currents are zero at the transition of S_1 or S_2 , zero switching for primary switches is confirmed. A single-phase AC-DC converter having parameters listed in Table 4.1 is also simulated in SABER®. In Fig. 4.10, the input voltage and filtered input current are in phase with each other, indicating that a unity power factor of operation is obtained. The calculated and simulated values for P_{io} , \bar{I}_i , \bar{I}_o and \bar{I}_{rpl} for both DC-DC and AC-DC converters match closely as listed in Table 4.2.



Figure 4.10: Simulation result: Single-phase AC-DC

Parameter	Simulation	Calculation	Simulation	Calculation
	(DC-DC) (DC-DC)		(AC-DC)	(AC-DC)
Pio	158.81 W	160 W	79.51 W	80 W
$\overline{I}_i(\text{RMS})$	10.12 A	10.06 A	7.27 A	7.35 A
$\overline{I}_o(\text{RMS})$	4.53 A	4.50 A	2.96 A	3.01 A
$\overline{I}_{rpl}(\text{RMS})$	4.46 A	4.43 A	2.93 A	2.97A

Table 4.2: Comparison of simulation and analytical values



Figure 4.11: Primary side converter in push-pull topology

4.3 Experimental setup

In this AC-DC PET, the primary side switches need to be four-quadrant. It can be constructed in two distinct ways: 1) Primary A: Using a diode bridge and one controlled switch as shown in Fig. 4.11(a). 2) Primary B: Using common-emitter or common-collector four-quadrant switches shown in Fig. 4.11(b). Both these cases are identical in the ideal-case and the primary switches are switched at zero current. However, a real circuit has device drops, conduction loss, dead-times etc. and the current may not cross zero at the switch transitions. Hence, a clamp circuit will be required for Primary A.

In Primary B, source based commutation will be possible. Consider the positive half cycle of v_i , switches S_{1b} and S_{2b} are kept on and S_{1a} and S_{2a} are modulated at 50% duty ratio at high-frequency. Consider the input voltage is in the positive half cycle, and winding 1 was conducting when S_{1a} is turned off and there is a dead-time between before switch S_{2a} is turned on. If the current in the transformer winding is slightly positive, the diode in the lower winding, d_{2a} and S_{2b} will conduct the current and S_{2a} will have ZVS turn-on. If the current is slightly negative, the diode in upper winding, d_{1a} and S_{1b} will be conducting so that S_{1a} as well as S_{2a} will be ZCS. There is no need to sense the direction of the current for the modulation of these primary side switches only the polarity of the voltage is necessary. In the negative half cycle, switches S_{1a} and S_{2a} are kept on and S_{1b} and S_{2b} are modulated at 50% duty ratio at high-frequency. A comparison of Primary A and Primary B is given below.

• Switching Loss In Primary A, the primary switches may not attain ZCS. In

Primary B, in each switching transition, one of the switches achieves either ZVS or ZCS. Also, only 2 switches are modulated at high frequency, the other two are modulated at line frequency at the zero crossing on the line voltage, where the switching loss is low.

- Conduction Loss: In Primary A, the primary switch has two diodes and one switch, while in Primary B, the path is only one diode and one switch. Hence conduction losses are less.
- Gate Driver: In Primary A, only two gate drive circuits are required. In Primary B, four gate drive circuits are required however since common-emitter configuration is used, the number of isolated power supplies required for these gate drivers is only two.

In this chapter, Primary A configuration is used to test the DC-DC converter and the single-phase AC-DC converter. The schematic for the experimental setup is in Fig. 4.12. The picture of the experimental setup is in Fig. 4.13. L_f, C_f are the input filter for this converter. v_{pps} is the programmable power supply for the single-phase AC-DC converter and a regulated DC power supply is used for the DC-DC converter. R_i and R_o are load resistors that are used to sink the power. Additional inductance is added on the secondary side of the transformer to limit the amount of power transferred. The details of the experimental setup are in Appendix B. The PWM pulses for all the power semiconductor devices are generated using an FPGA. The details of FPGA implementation are in Appendix E.1.

4.4 Experimental results: DC-DC converter

The experimental parameters are listed in Table. 4.3. The magnitude of V_o is set to 100V and V_i is varied to obtain different values of d. The converter is run for different values of d and δ and the results are summarized in Fig. 4.14(a) for positive values of δ and in Fig. 4.14(b) for negative values of δ . The input and output currents, i_i and i_o respectively that are marked in Fig. 4.12 are measured using a current probe and the average values of input power $P_i = v_i \times i_i$ and output power $P_o = v_o \times -i_o$ are calculated from the scope data. The expected power P_{io} equals $\delta d^2 \frac{V_o^2}{2Lf_s}$ from (4.10).



Figure 4.12: Single-phase AC-DC PET: Schematic of experimental setup



Figure 4.13: Experimental setup for DC-DC and single-phase AC-DC conversion

Table 4.3: Experimental parameters for DC-DC and single-phase AC-DC converter

V_o	100V			
f_s , f	$10 \rm kHz$, $60 \rm Hz$			
L_s	$80\mu H$			
turns-ratio	1:1:1			



Figure 4.14: DC-DC converter: Expected power P_{io} , input power $P_i(-+)$ and output power $P_o(-.*)$ for different values of d and δ

Bi-directional power flow capability of this converter has been demonstrated. For positive values of δ power flows from v_i to v_o and vice versa for negative δ . It is observed that for positive direction of power flow, P_i is higher than P_o and the opposite is true for negative power flow. This is because of the losses in the circuit.

The operating point with $V_i = 40$ V and $\delta = \pm 0.2$ is selected for further study. The switching pulse for S_1 and the converter voltage v_s are shown in Fig. 4.15. For $\delta = 0.2$, v_s is shifted to the right with respect to the pulses for S_1 . For $\delta = -0.2$, v_s is shifted to the left with respect to the pulses for S_1 . The input and output currents as well as the inductor current are in Fig. 4.16. The currents through S_1 and S_2 , are shown in Fig. 4.17. At the transition of S_1 and S_2 the currents through these switches are small but they are not ZCS.



Figure 4.15: DC-DC converter: (top to bottom) FPGA pulse for S_1 (C1:1V/div), voltage v_s (C3:50V/div) inductor current i_{L_s} (C4:4A/div) and output current i_o (C2:4A/div); time at 100 μ s/div.



Figure 4.16: DC-DC converter: (top to bottom) Input and output currents i_i (C1:4A/div) and i_o (C2:4A/div), voltage v_s (C3:50V/div) and inductor current i_{L_s} (C4:4A/div); time at 100 μ s/div.



Figure 4.17: DC-DC converter: (top to bottom) FPGA pulse for S_1 (C3:1V/div), Primary switch currents i_{S_1} (C1: 4A/div) and i_{S_2} (C2: 4A/div) and inductor current i_{L_s} (C4:4A/div); time at 100 μ s/div.

4.4.1 Effects of non-idealities in the circuit

In this topology, power flow is obtained by phase shifting the voltages across the effective leakage inductance of the transformer. This causes a change in the average value of the input and output currents. Power electronic devices have a finite resistance and voltage drop associated with them. Also, additional resistance/inductance is present in any circuit due to circuit layout. These non-idealities cause the converter currents and power flow to deviate from the ideal case. The converter is simulated in PLECS® for a non-ideal case with diode voltage drop of 2.5V, and IGBT device drop of 1.7V and a resistance of 1 Ω in series with the inductance of the transformer (L_s). These simulation results are compared with an ideal case as well as with the hardware results in Fig. 4.18 and Fig. 4.19. The results are compiled in Table 4.4.

The input and output voltage are assumed to be constant during a switching interval. However, this is not true as there is a ripple in the capacitor voltages. This causes additional error that may not result in ZCS of S_1 and S_2 .

The converter is controlled using an FPGA. The voltage sensors have some offset associated with them which causes an error in the calculated duty ratio d. Additionally, the devices have finite turn-on and turn-off times. This leads to an error in δ .

A dead-time is introduced between the upper and lower switches of the two level converter. During the dead-time the diodes form a path for the inductor current to



Figure 4.18: DC-DC converter: Comparison of simulation (top) and experimental (bottom) results of inductor current (i_{L_s}) .



Figure 4.19: DC-DC converter: Comparison of simulation (top) and experimental (bottom) results of DC current (i_o) .

	$\delta = 0.2$				$\delta = -0.2$	2
	ideal	non-ideal	experimental	ideal	non-ideal	experiment
P_o (W)	154.41	114.51	63.65	-212.21	-245.57	-252.56
\bar{I}_{L_s} (A)	6.34	6.38	5.70	6.39	6.61	5.51
\bar{I}_o (A)	3.74	3.37	3.04	4.30	4.61	4.33

Table 4.4: Experimental results: DC-DC converter



Figure 4.20: DC-DC converter: Effect of dead-time Input and output currents (top to bottom) i_i (C1:4A/div) and i_o (C2:4A/div), converter voltage v_s (C3:50V/div) and inductor current i_{L_s} (C4:4A/div); (left) time at 500 μ s/div and (right) time at 5 μ s/div.

flow. Fig. 4.20 shows the inductor current, and the converter voltage, v_s . The inductor current is slightly positive when switch S'_{o1} is turned off, hence, diode d_{o1} comes into conduction and the voltage $v_s = 100$ V. This causes the inductor current to reduce to zero. Once the current reaches zero, v_s the depends on the voltage balanced by the device capacitances. When S_{o1} is turned on, the voltage $v_s = 100$ V. Hence a glitch is observed in the voltage v_s .

4.5 Experimental results: Single-phase AC-DC converter

The experimental parameters are listed in Table. 4.3. The magnitude of V_o is set to 100V and \hat{V}_i is varied to obtain different values of m. The converter is run for different values



Figure 4.21: Single-phase AC-DC converter: Expected power P_{avg} , input power $P_i(-+)$ and output power $P_o(-.^*)$ for different values of m and δ .

of m and δ . The average input power, $P_i = v_i \times i_i$ and output power, $P_o = v_o \times -i_o$ are calculated from the scope data. The expected power P_{avg} equals $\delta m^2 \frac{V_o^2}{4Lf_s}$ from (4.24). The results are summarized in Fig. 4.21(a) for positive values of δ and in Fig. 4.21(b) for negative values of δ .

Bi-directional power flow capability of this converter has been demonstrated. It is observed that for positive direction of power flow, P_i is higher than P_o and the opposite is true for negative power flow. This is because of the losses in the circuit. The power flow in the converter has a quadratic relationship with respect to d and almost linear relationship with respect to δ .

In order to observe the switching waveforms of the converter $\hat{V}_i = 30\sqrt{2}$ and δ is selected to be ± 0.2 . The pulses for S_1 along with the phase shifted secondary side voltage, v_s are shown in Fig. 4.22. For $\delta = +0.2$, v_s is shifted to the right with respect to S_1 . For $\delta = -0.2$, v_s is shifted to the left with respect to S_1 . The input and output currents as well as the inductor current can be observed in Fig. 4.23. The current i_o in Fig 4.23(a) has a negative average value indicating power flow is from the AC to DC side. i_o has a positive average value in Fig 4.23(b) indicating power flow from the DC to AC side. The input voltage v_i along with the filtered input current, i_{L_f} are shown in Fig. 4.24. There is a phase shift between v_i and i_{L_f} because of the input filter. It can be seen that the non filtered converter current, i_i is in phase with the input voltage for



Figure 4.22: Single-phase AC-DC converter: (top to bottom) FPGA pulse for S_1 (C1:1V/div), output current i_o (C2:4A/div), v_s (C3:50V/div) and inductor current i_{L_s} (C4:4A/div); (left) time at 5ms/div and (right) time at 50 μ s/div.

a positive δ and it is 180⁰ out of phase with respect to v_i for a negative δ . The currents through switch S_1 and S_2 are shown in Fig. 4.25. When S_1 is on, the current through it equals the inductor current i_{L_s} . When S_1 is off, $i_{S_2} = i_{L_s}$. The Fourier spectrum of the input current, i_i when $\delta = 0.2$ and $\delta = -0.2$ are in Fig. 4.26(a) and Fig. 4.26(c) respectively. This current has harmonics at the fundamental of 60Hz (f) and at 20kHz $(2 \times f_s)$. The dc value of the output current has been removed and the Fourier transform of the resultant waveform when $\delta = 0.2$ and $\delta = -0.2$ are in Fig. 4.26(b) and Fig. 4.26(d) respectively. This current has components at 120Hz $(2 \times f)$ and at 20kHz $(2 \times f_s)$.

4.5.1 Effects of non-idealities in the circuit

It is observed that the power transferred in the experimental prototype is different from the expected values. The analysis in section 4.4.1 is true for this single-phase AC to DC converter as well. A simulation with device voltage drops is run in PLECS® and the simulation and experimental results for the input current, inductor current and output current are compared in Fig. 4.27, Fig. 4.28 and Fig. 4.29 respectively. The results of this comparison are compiled in Table. 4.5.

From Fig. 4.27, it is observed that the input voltage, v_i has a large ripple content at the switching frequency. In the ideal analysis, it is assumed that v_i has a constant value during a switching interval. However, in the experimental case, the input voltage



Figure 4.23: Single-phase AC-DC converter: (top to bottom) Input and output current i_i (C1:4A/div) and i_o (C2:4A/div), v_s (C3:50V/div) and inductor current i_{L_s} (C4:4A/div); (left) time at 5ms/div and (right) time at 50 μ s/div.



Figure 4.24: Single-phase AC-DC converter: (top to bottom) Input and output current i_i (C1:4A/div) and i_o (C2:4A/div), input voltage v_i (C3:50V/div) and filtered input current i_{L_f} (C4:4A/div); (left) time at 5ms/div and (right) time at 50 μ s/div.



Figure 4.25: Single-phase AC-DC converter: (top to bottom) Currents through the primary switches i_{S_1} , i_{S_2} (C1,C2:4A/div), FPGA pulse for S_1 (C3:1V/div) and inductor current i_{L_s} (C4:4A/div); (left) time at 5ms/div and (right) time at 50 μ s/div.

	$\delta = 0.2$				$\delta = -0.2$		
	ideal non-ideal experimental		ideal	non-ideal	experiment		
P_o	62.08	26.20	29.78	-185.67	-188.18	-141.25	
\bar{I}_{L_s}	5.39	5.35	4.47	5.42	5.20	4.43	
\bar{I}_o 2.94	2.73	2.27	3.96	3.90	3.17		

Table 4.5: Experimental results: Single-phase AC-DC converter

has some ripple it because of the ripple in the input current. This ripple in v_i has two effects; 1) In a switching interval, P_{io} given by (4.10) is proportional to the square of V_i , hence any perturbation in V_i will result in a change in the power transferred. 2) V_i is sensed only once every switching cycle. The duty ratio for the secondary side converter are calculated using this sensed input voltage. Hence, the applied duty ratio may not result in ZCS of the primary switches.

4.6 Conclusions and future work

In this chapter, a modulation technique for a DAB-based single-phase AC-DC PET is proposed that results in several beneficial features simultaneously. The salient features



Figure 4.26: Single-phase AC-DC converter: Fourier spectrum of input and output currents



Figure 4.27: AC-DC converter: Comparison of simulation (top) and experimental (bottom) results of input current (i_i) .



Figure 4.28: AC-DC converter: Comparison of simulation (top) and experimental (bottom) results of inductor current (i_{L_s}) .



Figure 4.29: AC-DC converter: Comparison of simulation (top) and experimental (bottom) results of DC current (i_o) .

are single-stage power conversion, bi-directional power flow, open-loop unity power factor of operation, soft switching and high power density.

This single-stage power conversion with high-frequency transformer isolation is an improvement over conventional multi-stage AC-DC converters with bulky low frequency transformers. The bi-directional property of this converter makes is especially attractive for distributed DC storage. The linear relationship between power and δ makes it easy to control.

By operating in the 'inner mode' with constant phase shift, open-loop unity power factor is obtained on the AC-side. Nearly Zero Current Switching (ZCS) on the low voltage (high-current) side will lead to reduction in switching losses and improved efficiency. The switching frequency can also be increased, thus reducing the size of the magnetic components considerably. However, this potential reduction in size needs to be evaluated in light of higher RMS currents that result to achieve ZCS.

The input current has harmonics at twice the switching frequency. This translates to a small size of filter on the AC side. The DC side current has a low frequency harmonic at twice the input fundamental frequency. Hence, a bigger filter capacitor is required for the DC side of this converter.

The converters for DC-DC power conversion and single-phase AC-DC power conversion have been tested extensively in simulation. Experimental prototypes have been built and tested. The device voltage drops, clamp circuit, dead-time, device turn-on and turn-off delays, sensing delays, ripple in input and output capacitor voltage are reasons for discrepancy in the experimental and the simulation results. The IGBTs used in the secondary side converter are rated for 1200V, 50A. Hence, the device non-idealities are significant at lower voltages, this leads to lower efficiency of the converter.

4.6.1 Future work

- Interleaving of the PET can be beneficial for reducing the ripple currents seen by the input and output side capacitors.
- Optimizing the filter design and the converter design to reduce the effects of parasitics on the operation of this PET.

Note: Parts of this chapter have been reprinted from [73] ©2012 IEEE Part of this work has been done in collaboration with Nathan and Kaushik.

Chapter 5

Three-phase AC-DC Power Electronic Transformer

High-frequency transformer isolated three-phase AC-DC converters with soft-switching features are discussed in this section. A typical multi-stage AC-DC converter with an active front-end for power factor correction (PFC) and a soft-switched DC-DC Dual Active Bridge converter to provided high-frequency isolation and regulated output voltage is shown in Fig. 5.1 [11, 41–43, 74, 75]. These two stages are coupled using a capacitor. Such multi-stage power conversion systems may require more than one controller, have more power electronic devices, lower efficiency and reduced reliability. The DC capacitors are further known to be unreliable especially under thermal stress. Therefore, there has been an increasing interest in single-stage converters for AC-DC power conversion.

Single-stage, HFT isolated, three-phase AC-DC converters with soft-switching (ZVS or ZCS) have been proposed in [45–47, 76]. These converters do not operate on the DAB principle, however they achieve single stage power conversion with ZVS/ZCS of



Figure 5.1: Multi-stage Power Electronic Transformer for AC to DC conversion

the switches. These converters however, lack bi-directional functionality.

Single-stage bi-directional converters that are soft-switched are proposed in [44] (resonant) and [48] (DAB). The converter in [48] is based on the DAB principle discussed in Chapter 4. It has features of ZCS and open loop power factor correction; however, the converters in [44, 48] require six four-quadrant switches on the AC side.

In this chapter, a three-phase bi-directional, HFT isolated, single-stage, DAB-based, AC-DC PET shown in Fig. 5.2 is analyzed. This converter has only two active switches on the AC-side. The transformers provide isolation and their leakage inductance is used for power transfer based on the DAB principle. This topology combines all the advantages of a HFT based system and a DAB-based system. Ideally, no clamp circuit is required for this converter. In Section 5.1, the topology and modulation technique are introduced. In Section 5.2, the converter is analyzed. In section 5.3, the analytical results are compared with the simulation results. In Section 5.4, closed-loop control of this converter is discussed. In Section 5.5, the experimental results on a laboratory prototype are presented.

5.1 Topology and modulation technique

The three-phase AC-DC PET is shown in Fig. 5.2. Balanced three-phase AC voltages given by (5.1) are applied to a bank of three three-winding transformers. The turns-ratio of one half of the primary winding to the secondary winding is 1 : n. The secondary side of the transformer is connected to a DC voltage source, V_{dc} through a two-level converter. In this topology, $V_{dc} > \sqrt{3}n\hat{V}_i$. All the switches on the DC side are two-quadrant. Each switch on the AC side is realized using a three-phase diode bridge and a single one-quadrant switch. S_1 and S_2 are switched in a complementary way, with 50% duty at a switching frequency $f_s\left(=\frac{1}{T_s}\right)$ where, $f_s \gg f$. Hence high-frequency AC voltages are applied to the transformer. Assuming the leakage inductance of the two primary windings are identical ($L_{p_1} = L_{p_2} = L_p$) and the magnetizing inductance is neglected, the primary side circuit and HFT referred to the secondary side can be represented by three balanced voltages v'_x (where, x denotes a, b or c phase) with equivalent transformer leakage inductances in series with it. v'_x is $+nv_x$ when S_1 is on and $-nv_x$ when S_2 is on. The per-phase equivalent leakage inductance referred to the



Figure 5.2: Three-phase AC-DC converter (PET): Topology



Figure 5.3: Two-level converter space vectors

secondary side is given by $L_{eq} = n^2 L_p + L_s$. The two-level converter is pulse-width modulated (PWM) to satisfy the following three conditions. 1) Over every $\frac{T_s}{2}$, the average voltage applied across each transformer equivalent leakage inductance must be zero. 2) The PWM voltages generated by the two-level converter are phase shifted with respect to the voltages on the secondary side of the transformer; hence, power is transferred by the Dual Active Bridge principle. 3) In this modulation, the phase shift that can be applied is limited such that at any point, the converter can be viewed to be in 'inner mode' of operation. In order to satisfy the first criterion, the duty ratios for the two-level converter are calculated using conventional Space Vector Modulation (SVM). The voltage space-vector diagram for a two-level converter is in Fig. 5.3. In this example, when S_1 is on, the input voltage space vector, \vec{V}^* given by (5.2) is in sector 1 and when S_2 is on, it is 180^0 out of phase, in sector 4. The duty ratios for vectors \vec{V}_1 and \vec{V}_2 are given by d_1 and d_2 respectively in (5.3). As the effective voltage applied across L_{eq} is zero every $\frac{T_s}{2}$ the phase currents of the transformer go to zero at the transition from S_1 to S_2 and vice-versa. Hence, Zero Current Switching (ZCS) is
achieved for these switches.

$$v_{a}(t) = \hat{V}_{i} \cos(\omega t)$$

$$v_{b}(t) = \hat{V}_{i} \cos\left(\omega t - \frac{2\pi}{3}\right)$$

$$v_{c}(t) = \hat{V}_{i} \cos\left(\omega t - \frac{4\pi}{3}\right) \qquad \cdots \text{ Where, } \omega = 2\pi f \qquad (5.1)$$

$$\vec{V}_i(t) = \vec{V}^*(t) = v'_a(t) + v'_b(t)e^{j\frac{2\pi}{3}} + v'_c(t)e^{j\frac{4\pi}{3}}$$
(5.2)

$$d_1 = \sqrt{3}d\sin\left(\frac{\pi}{3} - \alpha\right)$$

$$n\hat{V}_i$$

$$d_2 = \sqrt{3}d\sin(\alpha) \quad \cdots \text{ Where, } d = \frac{nV_i}{V_{dc}}$$
(5.3)

$$|\delta| \le \frac{1}{2} \left(1 - \sqrt{3}d \right) \quad \text{Hence, } d_{max} = \frac{1}{\sqrt{3}} \tag{5.4}$$

Without loss of generality, the converter waveforms are analyzed when \vec{V}^* is in sector 1 and S_1 is on. The secondary side equivalent transformer voltage for phase-a, v'_a and the two-level converter output voltage for phase-a, v_{AN} are shown in Fig. 5.4(a). A phase shift $\delta \frac{T_s}{2}$ is introduced between the primary and secondary PWM voltages of the transformer in order to obtain power transfer and v_{AN} is symmetric with respect to the phase shift $\frac{T_s}{4} + \delta \frac{T_s}{2}$. According to the third condition, the PWM pulses for the two-level inverter must be contained within $\frac{T_s}{2}$; thus, δ is limited by (5.4). When S_1 is on, v_{AN} is $0, \frac{2}{3}V_{dc}, \frac{1}{3}V_{dc}$ when vectors $\vec{V_0}, \vec{V_1}$ and $\vec{V_2}$ are applied respectively. The current through L_{eq} for phase-*a* is shown in Fig. 5.4(b). As the average value of v_{AN} equals the average value of v'_a over each half cycle, $I_{a_0} = I_{a_5} = 0$. This is true for phase-b and phase-c as well. When vector $\vec{V_0}$ is applied, the DC current, i_{dc} (Fig.5.4 (c)) is zero. When vector $\vec{V_1}$ is applied, $i_{dc} = i_{L_{eq},a}$ and when vector $\vec{V_2}$ is applied, $i_{dc} = i_{L_{eq},a} + i_{L_{eq},b}$. In order to analyze the currents, voltages and power the per-unit (pu) base values on the secondary side are chosen to be (5.5). The currents $i_{L_{eq},a}$ and i_{dc} are assumed to be piecewise linear over the switching cycle; hence, per-unit switching-cycle-average phase currents as well as the DC current of the converter can be calculated to be (5.6) and (5.7) respectively. It is clear from (5.6), if δ is a constant, unity power-factor is obtained on the AC side. The average power per switching cycle, P_{io} is given by (5.8). The power transferred is directly proportional to δ . The locus of the power that can be transferred for different values of d as a function of δ is plotted in Fig. 5.5. Maximum



Figure 5.4: Switching waveforms

power can be transferred (0.1164pu) when d is 0.3849 $(\frac{2}{3\sqrt{3}})$. When δ is positive, power is transferred from the AC side to the DC side, when δ is negative, the power transfer is in the opposite direction.

$$\left. \begin{array}{l} V_{base} = V_{dc} \\ I_{base} = \frac{V_{dc}}{2\pi f_s L_{eq}} \\ P_{base} = \frac{V_{dc}^2}{2\pi f_s L_{eq}} \end{array} \right\}$$
(5.5)

$$i_{a} = \pi d\delta \cos(\omega t)$$

$$\tilde{i}_{b} = \pi d\delta \cos(\omega t - \frac{2\pi}{3})$$

$$\tilde{i}_{c} = \pi d\delta \cos(\omega t - \frac{4\pi}{3})$$
(5.6)

$$\tilde{i}_{dc} = 1.5\pi d^2\delta \tag{5.7}$$

$$P_{io} = \frac{3d^2 V_{dc}^2 \delta}{4L_{eq} f_s}$$
$$= \frac{3d^2 \pi \delta}{2} p u$$
(5.8)

93



Figure 5.5: Three-phase AC-DC: P_{io} Vs δ

5.2 Analysis

Every $\frac{T_s}{2}$, the reference voltage space vector, \vec{V}^* is constructed using the two adjacent vectors that make up the sector along with zero vectors. If the zero vector is selected to be 0(000) in all sectors, the vector sequence 0-1-2-1-0 in odd sectors and 0-2-1-2-0 in even sectors will result in minimum number of switch transitions. If 7(111) is selected as the zero vector for all sectors, then switching sequence 7-2-1-2-7 in odd sectors and 7-1-2-1-7 in even sectors will result in minimum number of switch transitions. If the switching sequence is held constant, the zero vector that needs to be applied will change depending on the sector. This will lead to a switching from (111) to (000) every $\frac{T_s}{2}$ because, at that instant, the input voltage reference always switches from an odd sector to even sector or vice-versa. Since $f_s >> f$, in one switching interval, the input AC-voltages can be assumed to be constant, thus the slope of the currents in Fig. 5.4(b), (c) will be linear. The RMS value for the piece-wise linear DC current calculated over $\frac{T_s}{2}$ is denoted by $\bar{I}_{Ts/2}$, it is a function of voltages, d, α and δ . The value of $\bar{I}_{Ts/2}$ is calculated in all 6 sectors for switching sequence 0-1-2-1-0 and 0-2-1-2-0. The RMS current over a sector j, \bar{I}_j is calculated by (5.9). It can be shown that it is sufficient to calculate the value of

 \overline{I}_j in sector 1 and 2 only. Further, the RMS value over the low frequency fundamental is calculated by (5.10). A similar analysis is done for the transformer currents. The choice of switching sequence does not affect the RMS values for the currents and they all result in unity-power factor on the AC-side.

$$\overline{I}_{j} = \sqrt{\frac{3}{\pi} \int_{0}^{\frac{\pi}{3}} \overline{I}_{Ts/2}^{2} \,\mathrm{d}\alpha} \qquad j \in \{1, 2..6\}$$
(5.9)

$$\overline{I}_{1/f} = \sqrt{\frac{1}{6} \sum_{j=1}^{6} \overline{I}_{j}^{2}}$$
(5.10)

The RMS value of the DC-current \overline{I}_{dc} of this converter in per-unit is given by (5.11). The size of the DC-capacitor is a function of the RMS ripple current content of i_{dc} . This RMS ripple current in the DC capacitor, \overline{I}_{rpl} is calculated by subtracting the DC component from \overline{I}_{dc} . The values of \overline{I}_{rpl} given by (5.12) are plotted for different values of d as a function of δ in Fig. 5.6. For a certain value of d, the ripple content in the DC capacitor increases as the absolute value of phase shift, $|\delta|$ increases. The maximum value of \overline{I}_{rpl} is 0.137pu when d = 0.322.

$$\overline{I}_{dc} = K_2 [(\sqrt{3} \left(9d(-45+358d)+40 \left(29+360\delta^2\right)\right) - 1890d\pi)\pi d^3]^{\frac{1}{2}} \quad \text{Where, } K_2 = 0.013 \quad (5.11)$$

$$\overline{I}_{rpl} = K_2 [(\sqrt{3}(9d(-45+358d)+40(29+360\delta^2))$$

$$-270d(7+48\delta^2)\pi)\pi d^3]^{\frac{1}{2}}$$
(5.12)

$$\overline{I}_{L_{eq}} = K_1 [(-560\sqrt{3}d + 27d^2(3\sqrt{3} + 8\pi) + 96(\pi + 12\delta^2\pi))\pi d^2]^{\frac{1}{2}} \quad \text{Where, } K_1 = 0.021$$
(5.13)





Figure 5.6: Three-phase AC-DC: \overline{I}_{rpl} Vs δ

Figure 5.7: Three-phase AC-DC: P_t Vs P_{io}

$$\overline{V}_p = d/2$$
 $\overline{V}_s = \sqrt{\frac{2d}{\sqrt{3}\pi}}$ (5.14)

$$\bar{I}_{L_p} = \frac{I_{L_s}}{\sqrt{2}} \qquad \qquad \bar{I}_{L_s} = \bar{I}_{L_{eq}} \tag{5.15}$$

$$P_t = \frac{3}{2} \left[2\overline{V}_p \overline{I}_{L_p} + \overline{V}_s \overline{I}_{L_s} \right]$$
$$= \frac{3}{2} \left(\frac{d}{\sqrt{2}} + \sqrt{\frac{2d}{\sqrt{3}\pi}} \right) \overline{I}_{L_s}$$
(5.16)

The size of the high-frequency transformer is related to its VA rating which is given by P_t in (5.16). Hence, it is important to determine the RMS voltages and currents applied to the high-frequency transformer. The RMS value of $i_{L_{eq}}$ in per-unit as a function of d and δ is given by (5.13). The voltages generated by the two-level converter are applied at the secondary side of the HFT. Since the switching duty ratios as well as the DC voltage is known, the RMS voltages applied to the secondary side of the transformer, \overline{V}_s can be calculated in the same way as the RMS analysis for the currents. The transformer primary and secondary winding RMS voltages \overline{V}_p and \overline{V}_s are given by (5.14). As each primary winding conducts only for 50% of the time, \overline{I}_{L_p} is given by (5.15). P_t is plotted

\hat{V}_i	$20\sqrt{2}V$
V_{dc}	100V
$f_s\;,f$	$10 \rm kHz$, $60 \rm Hz$
L_s	$80\mu H$
turns-ratio	1:1:1
δ	± 0.2

Table 5.1: Simulation and experimental parameters for three-phase AC-DC converter

as a function of P_{io} for different values of d in Fig. 5.7. The transformer utilization factor (TUF) is defined as $\frac{P_{io}}{P_t}$. The highest value of TUF 68.8% is obtained for a power transfer of 0.116pu with d = 0.399. One drawback of this converter is that, when δ equals zero, the power transfer is zero, however there are RMS currents in the transformer. The worst case occurs when d = 0.577 and $P_t = 0.106$ pu.

5.3 Simulation results

The converter in Fig. 5.2 is simulated in PLECS®. All the switches and diodes in the circuit are considered to be ideal. The simulation parameters are listed in Table 5.1. The input current and voltage for phase-a (i_a , v_{an_i}) and the DC current (i_{dc}) are plotted for $\delta = 0.2$ and $\delta = -0.2$ in Fig. 5.8 and 5.9 respectively. It can be seen that bi-directional power flow is possible by changing the direction of phase shift, δ . The input currents have harmonics at twice the switching frequency and the output current has harmonics at six times the fundamental frequency in addition to harmonics at twice the switching frequency. The switching pulses for switch S_1 , the currents through the AC-side switches and the DC current are shown for a few switching cycles in Fig.5.10 and 5.11. It is observed that the switches S_1 and S_2 are switched at zero current.

The RMS values for transformer currents as well as DC current are measured for $\delta = \pm 0.2$, these values are compared with their analytical formulas in Table 5.2.



Figure 5.8: $\delta = 0.2$





Figure 5.10: $\delta = 0.2$



10.6



Table 5.2: Analytical and simulated values

Figure 5.12: DC load

Figure 5.13: Control loop

5.4Closed-loop control of three-phase AC-DC converter

In this section a controller is designed so as to regulate the DC voltage of the converter in Fig. 5.2. In this converter, the power transfer is directly proportional to the phase-shift δ and the equation for power transfer is given by (5.17). The DC side of the converter can be represented by the load capacitor and some load resistance R as shown in Fig. 5.12. R can be either positive or negative depending on the direction of power flow. The output DC current and DC voltage are related by (5.18). The controller can be designed by linearizing the circuit about an operating point. If there is a perturbation $\tilde{\delta}$ around the operating point δ , there is a perturbation in P, V_{dc} and I_{dc} which are represented by $\check{\tilde{p}}, v_{dc}^{\check{z}}$ and $i_{dc}^{\check{z}}$ respectively. The resulting perturbed power equation is given by (5.19). Neglecting the terms for $v_{dc}^{\check{i}}i_{dc}^{\check{i}}$, the resulting equation in frequency domain is given by (5.20). The transfer function relating V_{dc} and δ is in (5.21).

The control loop for the plant is shown in Fig. 5.13. A Proportional-Integral(PI)

Table 5.3: Simulation parameters: Closed-loop control

1	\hat{V}_i, V_{dc}	70V, 300V
	f_i, f_s	60Hz, 10 kHz
C	Z_{dc}, L_{eq}	$500\mu F, 100\mu H$

controller is designed for an operating point given in Table. 5.3. The cross-over frequency for the control is selected to be 50Hz.

$$P = V_{dc}I_{dc} = \frac{3\dot{V}_i^2\delta}{4L_{eq}f_s} \tag{5.17}$$

$$I_{dc}(s) = \frac{V_{dc}(s)(RCs+1)}{R}$$
(5.18)

$$P + \check{\tilde{p}} = (V_{dc} + v_{dc}^{\check{z}}) * (I_{dc} + i_{dc}^{\check{z}})$$

$$\check{\tilde{p}} = V_{dc}i\check{}_{dc} + I_{dc}\check{v}_{dc}$$
(5.19)

$$\frac{3\hat{V}_{i}^{2}\delta(s)}{4L_{eq}f_{s}} = V_{dc}\left[\frac{(RCs+1)V_{dc}(s)}{R}\right] + I_{dc}V_{dc}(s)$$
(5.20)

$$G_{ps}(s) = \frac{V_{dc}(s)}{\delta(s)} = \frac{3V_i^2}{4L_{eq}f_s} \frac{1}{\left[\frac{V_{dc}(RCs+1)}{R} + I_{dc}\right]}$$
(5.21)

The converter is simulated in using PLECS blockset in MATLAB[®]/Simulink[®] environment. The response of the controller to a step change in the reference value of V_{dc} from 300V to 330V at $\frac{2}{f_i}$ s is in Fig. 5.14. The load resistance is changed at $\frac{7}{f_i}$ s from R to $\frac{2}{3}R$ and it is seen that after an initial perturbation, the voltage is being controlled to the reference value. In order to study the controller response for bi-directional power flow, the output is represented as a current source that changes value so that the converter switches from feeding power to drawing power from the AC side at $6/f_i$ s. The DC voltage reference is set to 300V during this time. The DC voltage and currents are shown in Fig. 5.15, The unfiltered input voltage and current for phase-*a* are given in Fig. 5.16. It is clear that the input current is 180^0 out of phase before $3/f_i$ s and in phase with it after that, hence unity power factor is obtained as well as bi-directional power flow is possible under voltage control.



Figure 5.14: Voltage controller response



Figure 5.15: DC voltage and currents



Figure 5.16: AC current and voltage

5.5 Experimental results

The schematic of the experimental setup for the three-phase AC-DC converter is shown in Fig. 5.17. A photograph of the experimental setup is in Fig. 5.18. The three winding transformer are wound of ferrite cores. These transformers have unity turns-ratio, leakage inductance of approximately 15μ H and magnetizing inductance of 50mH. The leakage inductance of the transformer is very small, hence, in order to limit the power flow in this converter, additional inductors, L_A, L_B and L_C are connected on the secondary side of the transformers as shown in Fig.5.17. Since programmable power supplies are used in this experiment, the resistors R_i and R_{dc} are included to sink the power that is transferred by this AC-DC converter. An LC ($L_f=0.5$ mH, $C_f=20\mu$ F) filter is used on the AC side. A clamp circuit composed of a resistor and capacitor is present on the AC side. It provides a path for currents to flow during the dead-time between switches S_1 and S_2 . More details on the experimental setup are in Appendix B. The input voltages, v_{ab}, v_{ac} and v_{dc} are sampled at 10kHz using 12-bit ADCs. Using these sensed voltages, the FPGA computes the duty ratios and generates PWM pulses for all the power electronic switches. The details of the FPGA implementation are in Appendix E.2.

The converter is tested at different values of d and δ for parameters listed in Table. 5.1. The magnitude of V_o for all these cases is 100V and \hat{V}_i is varied to obtain different values of d. In order to compare the analytical values and experimental values, it is assumed that the DC voltage, V_{dc} is constant at 100V. The RMS value of the input voltage for phase-a (\overline{V}_a) is calculated from the scope data. The value of $d = \frac{\sqrt{2} \times \overline{V}_a}{V_{dc}}$. The average input power for phase-a is calculated by taking the average of the instantaneous product of the input voltage and unfiltered input current. The average value of the DC current is given by I_{dc} . The output power, $P_{dc} = V_{dc} \times -I_{dc}$. The results are summarized in Fig. 5.19(a) for positive values of δ and in Fig. 5.19(b) for negative values of δ . There is some error in the expected power, P_{io} (5.8) and the measured AC and DC power. This is mainly due to the parasitics such as device voltage drops, resistance and dead-time. An operating point with $V_{dc} = 100V$, $\hat{V}_i = 20\sqrt{2}$ and $\delta = 0.2$ and $\delta = -0.2$ is selected for further study. The filtered and unfiltered input current for phase-a is in phase with the input voltage. Hence, unity power factor is obtained. When



Figure 5.17: Schematic of experimental setup



Figure 5.18: Photograph of experimental setup



Figure 5.19: Three-phase AC-DC converter: Expected power $P_{io}(-o)$, input power $P_i(-+)$ and output power $P_{dc}(-,*)$ for different values of d and δ

Table 5.4: Experimental and analytical results of P_{io}

	P_i (Watt)	P_{dc} (Watt)	P_{io} (Watt) (Analytical)
$\delta = 0.2$	102.72	-54.42	154.37
$\delta = -0.2$	-101.28	154.39	-161.76

Table 5.5: Experimental and analytical results of RMS currents				
	$\overline{I}_{L_{eq}}(\text{experiment})$	$\overline{I}_{L_{eq}}(\text{analytical})$	$\overline{I}_{dc}(\text{experiment})$	\overline{I}_{dc} (analytical)
$\delta = 0.2$	2.84A	3.23A	2.08A	2.82A
$\delta = -0.2$	2.60A	3.29A	2.69A	2.90A

 δ is negative, the input currents are 180^0 out of phase with the input voltage. Hence, power is flowing from DC to AC side. The average value of the DC current is positive when power flow is from the DC to AC side and it is negative for a positive value of δ . The experimental and analytical powers at the input and output of this converter are summarized in Table 5.4. The transformer secondary voltage and the converter voltage are shown in Fig. 5.21. For a positive δ , the two-level converter voltages are shifted to the right with respect to the high-frequency voltages on the secondary side of the transformer. When δ is negative, the two-level converter voltages are shifted to the left. The currents through the primary switches S_1 and S_2 are in Fig. 5.22. The switches, S_1 and S_2 are not critically switched at zero current. This is because of device voltage drops, circuit resistance, dead-time as well as discrepancy in duty ratio calculations. However, it must be noted that the current at the instant of switching is very small.

The RMS currents from the experimental data are compared with the analytical values in Table 5.5.

The Fourier spectrum of the unfiltered input current, i_a when $\delta = 0.2$ and $\delta = -0.2$ are in Fig. 5.23(a) and Fig. 5.23(c) respectively. This current has harmonics at the fundamental of 60Hz and at 20kHz (2 × f_s). The DC value of the output current has been removed and the Fourier transform of the resultant waveform when $\delta = 0.2$ and $\delta = -0.2$ are in Fig. 5.23(b) and Fig. 5.23(d) respectively. This current has components at 120Hz (2 × f_i), 360Hz (6 × f_i) and at 20kHz (2 × f_s).



Figure 5.20: Left: (top) Filtered input current i_{af} and the converter AC current i_a (2A/div) for phase-*a* along with the input voltage v_{an_i} (10V/div); (bottom) DC current i_{dc} (2A/div) time is 5ms/div. Right: zoomed in waveforms at 20 μ s/div



Figure 5.21: Open circuit voltages: (top) Input voltage v_{ab} , transformer secondary voltage $v_{A'B'}$ and output voltage of the converter between phase A and B (50V/div). (bottom) Zoomed in section of $v_{A'B'}$ and v_{AB} at 50 μ s/div



Figure 5.22: (top) Primary switch currents i_{S_1} , i_{S_2} (2A/div); (bottom) input voltage v_{an_i} and the filtered input current i_{L_f} for phase-*a* at 50 μ s/div.



Figure 5.23: Fourier spectrum of input and output currents

5.6 Conclusions and future work

In this research, a single-stage three-phase AC-DC PET with only two active switches on the AC-side has been analyzed in detail. This topology has all the benefits of a highfrequency transformer system:- voltage transformation ratio, isolation along with high power density. The DAB-based control provides all the benefits of DAB-converters:- soft switching, use of the transformer leakage inductance for energy transfer and compact size. The currents in this converter have high ripple percentage, however this ripple content is at $2 \times f_s$; hence, the size of the filters will be small. The high RMS values for the transformer have to be considered against the benefit of soft-switching.

In an ideal case, the switches S_1 and S_2 are ZCS therefore a clamp circuit is not required for these switches. This is not the case when the device voltage drops and circuit parasitics are taken into account. Hence, a clamp circuit is required on the primary side for the switches S_1 and S_2 . It should be noted that the current through the switches is very small at the switching instant.

The power transfer for this converter is directly proportional to the phase shift δ . Bi-directional power flow is possible by changing the sign of δ . The design of the voltage controller for this converter is given in this chapter. The simulation results show that the DC voltage can be regulated under small load disturbances.

An experimental prototype of this converter has been built and tested and the experimental results verify the advantages of this converter topology.

5.6.1 Future Work

• The operation of this converter under unbalanced voltages or as a single phase converter.

Note: Parts of this chapter have been reprinted from [77] ©2012 IEEE Part of this work has been done in collaboration with Kaushik.

Chapter 6

Conclusion

In this research, PETs have been proposed for single-stage AC to AC and AC to DC power conversion with bi-directional power flow capabilities. The novelty of these converters is that they require only two controlled power electronic switches on the primary side. These primary switches operate at a constant frequency and at 50% duty ratio making them easy to control.

The features of the power electronic transformer for AC-AC conversion are listed below,

- Single-stage power conversion with high-frequency transformer isolation and bidirectional power flow capability.
- Variable frequency and amplitude pulse width modulated voltage generation.
- Variable power factor correction.
- Zero current switching (ZCS) for the primary side converter switches.
- Compact size and easy control technique that can be implemented on a single FPGA.

The features of the power electronic transformer for AC-DC conversion are summarized below,

• Single-stage power conversion with high-frequency transformer isolation and bidirectional power flow capability.

- Linear power relationship for easy control implementation.
- Unity power factor under open-loop control.

These two PETs without any intermediate storage capacitors and with their use of a high-frequency transformer link provide an efficient, reliable and high-power density solution for AC to AC and AC to DC power conversion.

References

- W. M. Murray, "Power conversion circuit having a high frequency link," US Patent Application US3.517.300, April 1968.
- [2] E. Ronan, S. Sudhoff, S. Glover, and D. Galloway, "A power electronic-based distribution transformer," *Power Delivery*, *IEEE Transactions on*, vol. 17, no. 2, pp. 537–543, apr 2002.
- [3] W. Reass, D. Baca, R. Gribble, D. Anderson, J. Przybyla, R. Richardson, J. Clare, M. Bland, and P. Wheeler, "High-frequency multimegawatt polyphase resonant power conditioning," *Plasma Science*, *IEEE Transactions on*, vol. 33, no. 4, pp. 1210 – 1219, aug. 2005.
- [4] J. Brooks, "Solid state transformer concept development," Final report Naval Construction Battalion Center, 1980.
- [5] M. Das, C. Capell, D. Grider, R. Raju, M. Schutten, J. Nasadoski, S. Leslie, J. Ostop, and A. Hefner, "10 kv, 120 a sic half h-bridge power mosfet modules suitable for high frequency, medium voltage applications," in *Energy Conversion Congress and Exposition (ECCE)*, 2011 IEEE, sept. 2011, pp. 2689–2692.
- [6] R. K. Gupta, G. F. Castelino, K. K. Mohapatra, and N. Mohan, "A novel integrated three-phase, switched multi-winding power electronic transformer converter for wind power generation system," in *Proc. Industrial Electronics Society, 2009. IECON'09. The 35th Annual Conference of the IEEE*, Porto, Portugal, Nov. 2009.

- [7] B. Kramer, S. Chakraborty, and B. Kroposki, "A review of plug-in vehicles and vehicle-to-grid capability," in *Industrial Electronics*, 2008. IECON 2008. 34th Annual Conference of IEEE, 2008, pp. 2278–2283.
- [8] T. Yiyun, L. Can, C. Lin, and L. Lin, "Research on vehicle-to-grid technology," in Computer Distributed Control and Intelligent Environmental Monitoring (CD-CIEM), 2011 International Conference on, feb. 2011, pp. 1013–1016.
- [9] W. Kempton and S. Letendre, "Electric vehicles as a new power source for electric utilities," in *Transportation Research*, 1997.
- [10] S. Vaishnav and H. Krishnaswami, "Single-stage isolated bi-directional converter topology using high frequency ac link for charging and v2g applications of phev," in Vehicle Power and Propulsion Conference (VPPC), 2011 IEEE, 2011, pp. 1–4.
- [11] H. Krishnaswami and V. Ramanarayanan, "Control of high-frequency ac link electronic transformer," *Electric Power Applications, IEE Proceedings* -, vol. 152, no. 3, pp. 509–516, 6 May 2005.
- [12] X. She, R. Burgos, G. Wang, F. Wang, and A. Huang, "Review of solid state transformer in the distribution system: From components to field application," in *Energy Conversion Congress and Exposition (ECCE)*, 2012 IEEE, 2012, pp. 4077–4084.
- [13] S. Falcones, X. Mao, and R. Ayyanar, "Topology comparison for solid state transformer implementation," in *Power and Energy Society General Meeting*, 2010 *IEEE*, 2010, pp. 1–8.
- [14] K. Harada, F. Anan, K. Yamasaki, M. Jinno, Y. Kawata, and T. Nakashima, "Intelligent transformer," in *Power Electronics Specialists Conference*, 1996. PESC '96 Record., 27th Annual IEEE, vol. 2, jun 1996, pp. 1337 –1341 vol.2.
- [15] M. Kang, P. Enjeti, and I. Pitel, "Analysis and design of electronic transformers for electric power distribution system," *Power Electronics, IEEE Transactions on*, vol. 14, no. 6, pp. 1133 –1141, nov 1999.

- [16] D. Chen and J. Liu, "The uni-polarity phase-shifted controlled voltage mode ac-ac converters with high frequency ac link," *Power Electronics, IEEE Transactions on*, vol. 21, no. 4, pp. 899 – 905, july 2006.
- [17] D. Chen, L. Li, J. Liu, S. Lin, and C. Song, "Novel current mode ac/ac converters with high frequency ac link," in *Power Electronics Specialists Conference*, 2005. *PESC '05. IEEE 36th*, june 2005, pp. 39–44.
- [18] H. Qin and J. Kimball, "A comparative efficiency study of silicon-based solid state transformers," in *Energy Conversion Congress and Exposition (ECCE)*, 2010 *IEEE*, sept. 2010, pp. 1458 –1463.
- [19] H. Krishnamoorthy, P. Enjeti, I. Pitel, and J. Hawke, "New medium-voltage adjustable speed drive (asd) topologies with medium-frequency transformer isolation," in *Power Electronics and Motion Control Conference (IPEMC)*, 2012 7th International, vol. 2, june 2012, pp. 814-819.
- [20] J. Shah, B. Wollenberg, and N. Mohan, "Decentralized power flow control for a smart micro-grid," in *Power and Energy Society General Meeting*, 2011 IEEE, 2011, pp. 1–6.
- [21] T. Zhao, L. Yang, J. Wang, and A. Huang, "270 kva solid state transformer based on 10 kv sic power devices," in *Electric Ship Technologies Symposium*, 2007. ESTS '07. IEEE, may 2007, pp. 145 –149.
- [22] H. Iman-Eini, J. Schanen, S. Farhangi, J. Barbaroux, and J. Keradec, "A power electronic based transformer for feeding sensitive loads," in *Power Electronics Specialists Conference*, 2008. PESC 2008. IEEE, june 2008, pp. 2549 –2555.
- [23] C. Ling, B. Ge, D. Bi, and Q. Ma, "An effective power electronic transformer applied to distribution system," in *Electrical Machines and Systems (ICEMS)*, 2011 International Conference on, aug. 2011, pp. 1–6.
- [24] M. Kim, A. Yoo, and S.-K. Sul, "A new universal isolated converter for grid connection," *Industry Applications, IEEE Transactions on*, vol. 48, no. 2, pp. 685–696, 2012.

- [25] S. Bifaretti, P. Zanchetta, A. Watson, L. Tarisciotti, and J. Clare, "Advanced power electronic conversion and control system for universal and flexible power management," *Smart Grid, IEEE Transactions on*, vol. 2, no. 2, pp. 231–243, 2011.
- [26] G. Wang, S. Baek, J. Elliott, A. Kadavelugu, F. Wang, X. She, S. Dutta, Y. Liu, T. Zhao, W. Yao, R. Gould, S. Bhattacharya, and A. Huang, "Design and hardware implementation of gen-1 silicon based solid state transformer," in *Applied Power Electronics Conference and Exposition (APEC), 2011 Twenty-Sixth Annual IEEE*, 2011, pp. 1344–1349.
- [27] K. Mohapatra and N. Mohan, "Matrix converter fed open-ended power electronic transformer for power system application," in *Power and Energy Society General Meeting - Conversion and Delivery of Electrical Energy in the 21st Century, 2008 IEEE*, july 2008, pp. 1–6.
- [28] K. Basu and N. Mohan, "A power electronic transformer for three phase pwm ac/ac drive with loss less commutation and common-mode voltage suppression," in *IECON 2010 - 36th Annual Conference on IEEE Industrial Electronics Society*, nov. 2010, pp. 315 – 320.
- [29] S. Nath, K. Mohapatra, K. Basu, and N. Mohan, "Source based commutation in matrix converter fed power electronic transformer for power systems application," in *Power Electronics Electrical Drives Automation and Motion (SPEEDAM)*, 2010 International Symposium on, june 2010, pp. 1682 –1686.
- [30] H. Cha and P. Enjeti, "A three-phase ac/ac high-frequency link matrix converter for vscf applications," in *Power Electronics Specialist Conference*, 2003. PESC '03. 2003 IEEE 34th Annual, vol. 4, june 2003, pp. 1971 – 1976 vol.4.
- [31] K. Basu, A. Umarikar, K. Mohapatra, and N. Mohan, "High-frequency transformerlink three-level inverter drive with common-mode voltage elimination," in *Power Electronics Specialists Conference*, 2008. PESC 2008. IEEE, june 2008, pp. 4413 -4418.

- [32] J. Ai-juan, L. Hang-tian, and L. Shao-long, "A new matrix type three-phase fourwire power electronic transformer," in *Power Electronics Specialists Conference*, 2006. PESC '06. 37th IEEE, june 2006, pp. 1–6.
- [33] R. Gupta, K. Mohapatra, and N. Mohan, "Novel topologies of power electronic transformers with reduced switch-count," in *Grand Challenges in Modeling and Simulation (GCMS'09)*, July 2009.
- [34] —, "A novel three-phase switched multi-winding power electronic transformer," in *Energy Conversion Congress and Exposition*, 2009. ECCE 2009. IEEE, sept. 2009, pp. 2696 –2703.
- [35] G. Castelino, K. Basu, R. Gupta, and N. Mohan, "Power electronic transformer with reduced number of switches: Analysis of input clamp circuit and a modulation strategy to eliminate input snubber requirement," in *IECON-USA*, 2010.
- [36] G. Castelino, K. Basu, and N. Mohan, "Power electronic transformer with reduced number of switches: Analysis of clamp circuit for leakage energy commutation," in *Power Electronics, Drives and Energy Systems (PEDES) 2010 Power India, 2010 Joint International Conference on*, dec. 2010, pp. 1–8.
- [37] R. Gupta, K. Mohapatra, N. Mohan, G. Castelino, K. Basu, and N. Weise, "Soft switching power electronic transformer," U.S. Patent application number: 20110007534, January 2011.
- [38] B. Ozpineci and B. Bose, "Soft-switched performance-enhanced high frequency nonresonant link phase-controlled converter for ac motor drive," in *Industrial Electronics Society*, 1998. IECON '98. Proceedings of the 24th Annual Conference of the IEEE, vol. 2, aug-4 sep 1998, pp. 733 –739 vol.2.
- [39] T. Kawabata, K. Honjo, N. Sashida, K. Sanada, and M. Koyama, "High frequency link dc/ac converter with pwm cycloconverter," in *Industry Applications Society* Annual Meeting, 1990., Conference Record of the 1990 IEEE, oct. 1990, pp. 1119 -1124 vol.2.

- [40] R. Garcia-Gil, J. Espi, E. Sanchis-Kilders, V. Esteve, J. Jordan, and E. Maset, "A dsp-controlled four-quadrant ac-dc matrix converter with high-frequency isolation," in Applied Power Electronics Conference and Exposition, 2004. APEC '04. Nineteenth Annual IEEE, vol. 2, 2004, pp. 1194 – 1199 vol.2.
- [41] J. Zhao, J. Jiang, and X. Yang, "Ac-dc-dc isolated converter with bidirectional power flow capability," *Power Electronics*, *IET*, vol. 3, no. 4, pp. 472 –479, july 2010.
- [42] M.-S. Huang, P.-Y. Yeh, J.-R. Huang, and C.-H. Liao, "Novel bi-directional acdc converter for electrical vehicle battery testing," in *IECON 2011 - 37th Annual Conference on IEEE Industrial Electronics Society*, nov. 2011, pp. 1480 –1485.
- [43] D. Segaran, D. Holmes, and B. McGrath, "High-performance bi-directional ac-dc converters for phev with minimised dc bus capacitance," in *IECON 2011 - 37th Annual Conference on IEEE Industrial Electronics Society*, nov. 2011, pp. 3620 –3625.
- [44] S. Norrga, S. Meier, and S. Ostlund, "A three-phase soft-switched isolated ac/dc converter without auxiliary circuit," *Industry Applications, IEEE Transactions on*, vol. 44, no. 3, pp. 836–844, may-june 2008.
- [45] D. Wijeratne and G. Moschopoulos, "A simple three-phase single-stage ac-dc zvzcs full bridge converter," in *Energy Conversion Congress and Exposition*, 2009. ECCE 2009. IEEE, sept. 2009, pp. 1220 –1227.
- [46] V. Vlatkovic, D. Borojevic, and F. Lee, "A zero-voltage switched, three-phase isolated pwm buck rectifier," *Power Electronics, IEEE Transactions on*, vol. 10, no. 2, pp. 148–157, mar 1995.
- [47] Y. Panov, F. Lee, and J. Cho, "A new three-phase ac-dc zero-voltage-switching isolated converter operating in dcm," in *Industry Applications Conference*, 1995. *Thirtieth IAS Annual Meeting, IAS '95., Conference Record of the 1995 IEEE*, vol. 3, oct 1995, pp. 2602 –2609 vol.3.

- [48] N. Weise, K. Basu, and N. Mohan, "A novel modulation strategy for a three-phase ac-dc dual active bridge with input power factor correction and simplified control," in *POWERENG*, 2011, 2011, pp. 1–8.
- [49] S. Norrga, "Experimental study of a soft-switched isolated bidirectional ac ndash;dc converter without auxiliary circuit," *Power Electronics, IEEE Transactions on*, vol. 21, no. 6, pp. 1580 –1587, nov. 2006.
- [50] H.-S. Kim, M.-H. Ryu, J.-W. Baek, and J.-H. Jung, "High-efficiency isolated bidirectional ac x2013;dc converter for a dc distribution system," *Power Electronics*, *IEEE Transactions on*, vol. 28, no. 4, pp. 1642 –1654, april 2013.
- [51] R. De Doncker, D. Divan, and M. Kheraluwala, "A three-phase soft-switched high power density dc/dc converter for high power applications," in *Industry Applications Society Annual Meeting*, 1988., Conference Record of the 1988 IEEE, Oct. 1988, pp. 796 –805 vol.1.
- [52] M. Kheraluwala, R. Gasgoigne, D. Divan, and E. Bauman, "Performance characterization of a high power dual active bridge dc/dc converter," in *Industry Applications Society Annual Meeting*, 1990., Conference Record of the 1990 IEEE, Oct. 1990, pp. 1267 –1273 vol.2.
- [53] K. Basu, R. Gupta, S. Nath, G. Castelino, K. Mohapatra, and N. Mohan, "Research in matrix-converter based three-phase power-electronic transformers," in *IPEC-Sapporo*, 2010.
- [54] S. Nath, K. Mohapatra, and N. Mohan, "Output voltage regulation in matrix converter fed power electronic transformer for power systems application in electric ship," in *Electric Ship Technologies Symposium*, 2009. ESTS 2009. IEEE, april 2009, pp. 203–206.
- [55] A. Shahani, K. Basu, and N. Mohan, "A power electronic transformer based on indirect matrix converter for pwm ac drive with lossless commutation of leakage energy," in *Power Electronics, Machines and Drives (PEMD 2012), 6th IET International Conference on*, march 2012, pp. 1–6.

- [56] P. Wheeler, J. Rodriguez, J. Clare, L. Empringham, and A. Weinstein, "Matrix converters: a technology review," *Industrial Electronics, IEEE Transactions on*, vol. 49, no. 2, pp. 276 –288, apr 2002.
- [57] R. Gupta, K. Mohapatra, A. Somani, and N. Mohan, "Direct-matrix-converter based drive for a three-phase open-end-winding ac machine with advanced features," *Industrial Electronics, IEEE Transactions on*, vol. PP, no. 99, pp. 1–1, 2010.
- [58] M. Milanovic and B. Dobaj, "Unity input displacement factor correction principle for direct ac to ac matrix converters based on modulation strategy," *Circuits and Systems I: Fundamental Theory and Applications, IEEE Transactions on*, vol. 47, no. 2, pp. 221–230, feb 2000.
- [59] N. Burany, "Safe control of four-quadrant switches," in Industry Applications Society Annual Meeting, 1989., Conference Record of the 1989 IEEE, Oct., pp. 1190– 1194 vol.1.
- [60] G. Castelino and N. Mohan, "Modulation and commutation of matrix converter based power electronic transformer using a single fpga," in accepted for publication IECON 2013.
- [61] N. Schibli, "Dc-dc converters for two-quadrant operation with controlled output voltage," in *EPFL*, 1999.
- [62] F. Krismer, J. Biela, and J. Kolar, "A comparative evaluation of isolated bidirectional dc/dc converters with wide input and output voltage range," in *Industry Applications Conference*, 2005. Fourtieth IAS Annual Meeting. Conference Record of the 2005, vol. 1, oct 2005, pp. 599 – 606 Vol. 1.
- [63] F. Krismer, S. Round, and J. Kolar, "Performance optimization of a high current dual active bridge with a wide operating voltage range," in *Power Electronics* Specialists Conference, 2006. PESC '06. 37th IEEE, june 2006, pp. 1–7.
- [64] G. Ortiz, J. Biela, D. Bortis, and J. Kolar, "1 megawatt, 20 khz, isolated, bidirectional 12kv to 1.2kv dc-dc converter for renewable energy applications," in *Power Electronics Conference (IPEC), 2010 International*, june 2010, pp. 3212 –3219.

- [65] H. Tao, A. Kotsopoulos, J. Duarte, and M. Hendrix, "A soft-switched three-port bidirectional converter for fuel cell and supercapacitor applications," in *Power Electronics Specialists Conference, 2005. PESC '05. IEEE 36th*, june 2005, pp. 2487 –2493.
- [66] H. Tao, A. Kotsopoulos, and J. Duarte, "Transformer-coupled multiport zvs bidirectional dc-dc converter with wide input range," *Power Electronics, IEEE Transactions on*, vol. 23, no. 2, pp. 771–781, march 2008.
- [67] A. K. Jain and R. Ayyanar, "Pwm control of dual active bridge: Comprehensive analysis and experimental verification," *Power Electronics, IEEE Transactions on*, vol. PP, no. 99, p. 1, 2010.
- [68] Y. Wang, S. de Haan, and J. Ferreira, "Optimal operating ranges of three modulation methods in dual active bridge converters," in *Power Electronics and Motion Control Conference, 2009. IPEMC '09. IEEE 6th International*, may 2009, pp. 1397–1401.
- [69] M. Kheraluwala and R. De Doncker, "Single phase unity power factor control for dual active bridge converter," in *Industry Applications Society Annual Meeting*, 1993., Conference Record of the 1993 IEEE, Oct. 1993, pp. 909 –916 vol.2.
- [70] K. Vangen, T. Melaa, and A. Adnanes, "Soft-switched high-frequency, high power dc/ac converter with igbt," in *Power Electronics Specialists Conference*, 1992. *PESC '92 Record.*, 23rd Annual IEEE, jun-3 jul 1992, pp. 26-33 vol.1.
- [71] K. Vangen, T. Melaa, S. Bergsmark, and R. Nilsen, "Efficient high-frequency softswitched power converter with signal processor control," in *Telecommunications Energy Conference*, 1991. INTELEC '91., 13th International, nov 1991, pp. 631 -639.
- [72] N. Weise, K. Mohapatra, and N. Mohan, "Universal utility interface for plug-in hybrid electric vehicles with vehicle-to-grid functionality," in *Power and Energy Society General Meeting*, 2010 IEEE, july 2010, pp. 1–8.
- [73] G. Castelino, K. Basu, N. Weise, and N. Mohan, "A bi-directional, isolated, singlestage, dab-based ac-dc converter with open-loop power factor correction and other

advanced features," in Industrial Technology (ICIT), 2012 IEEE International Conference on, 2012, pp. 938–943.

- [74] B. Reese, M. Schupbach, A. Lostetter, B. Rowden, R. Saunders, and J. Balda, "High voltage, high power density bi-directional multi-level converters utilizing silicon and silicon carbide (sic) switches," in *Applied Power Electronics Conference* and Exposition, 2008. APEC 2008. Twenty-Third Annual IEEE, feb. 2008, pp. 252 -258.
- [75] A. Yoo, K. Kim, and S.-K. Sul, "Analysis of high-frequency power conversion for series-connected universal link ac/dc power converter," in *Power Electronics and ECCE Asia (ICPE ECCE), 2011 IEEE 8th International Conference on*, 2011, pp. 184–191.
- [76] F.-S. Hamdad and A. K. S. Bhat, "A novel soft-switching high-frequency transformer isolated three-phase ac-to-dc converter with low harmonic distortion," *Power Electronics, IEEE Transactions on*, vol. 19, no. 1, pp. 35–45, 2004.
- [77] G. Castelino, K. Basu, and N. Mohan, "A novel three-phase bi-directional, isolated, single-stage, dab-based ac-dc converter with open-loop power factor correction," in *Power Electronics, Drives and Energy Systems (PEDES), 2012 IEEE International Conference on*, 2012, pp. 1–6.

Appendix A

Acronyms

PET	Power Electronic Transformer
HFT	High-Frequency Transformer
PWM	Pulse Width Modulation
MC	Matrix Converter
PHEVs	Plug-in Hybrid Electric Vehicles
PSM	Phase Shift Modulation
DAB	Dual Active Bridge
CCW	Counter-Clockwise synchronously rotating vectors used for
	modulation
CW	Clockwise synchronously rotating vectors used for modula-
	tion
CCW+CW	Counter-Clockwise rotating vectors used for modulation for
	one $\operatorname{cycle}(T_s)$ and Clockwise vectors used for modulation in
	the next cycle
RMS	Root mean square
THD	Total Harmonic Distortion
TWD	Total Waveform Distortion
ZCS	Zero Current Switching
ZVS	Zero Voltage Switching

Appendix B

Experimental Setup

B.1 Matrix converter board

The matrix converter (MC) is composed of Microsemi APTGF50TDU120PG IGBT modules that are rated for 1200V and 50A. Each module contains six IGBTs connected in common-emitter configuration as shown in Fig. B.1. A picture of the MC board is in Fig. B.2.

B.1.1 Clamp circuits

Three clamp circuits are part of the AC-AC PET setup. The primary clamp (V_{pri} in Fig. 2.10) provides a path for the transformer currents to flow during the dead-time between switches S_1 and S_2 . Additionally, it provides protection in case of a fault in the switching of S_1 or S_2 . The voltage of this clamp circuit is greater than $2\sqrt{3}\hat{V}_{an_i}$ so that it does not come into the circuit under normal operation. Clamp circuits are present at the input and output terminal of the matrix converter and are necessary for commutation of currents when the MC is switched. They are also required for protection during a fault in the MC. The diodes in the clamp circuit are fast acting diodes (IXYS DSEE30-12A) rated for 600V and 30A. These clamp circuits are terminated in an RC network with C=47 μ F and R depending on the operating point.



Figure B.1: Matrix converter board schematic



Figure B.2: Matrix converter board



Figure B.3: Matrix converter board configured as a two-level inverter



Figure B.4: Primary circuit: PET for AC-AC conversion

B.2 Two-level converter

The two-level converter is configured from the matrix converter board as shown in Fig. B.3. Some of the IGBTs switches have been shorted out to reduce the device voltage drop associated with these devices in the conduction path.

B.3 Primary circuit

1. AC-AC PET: The primary side circuit for this converter is in Fig. B.4. The primary side switches need to block a maximum voltage of $2\sqrt{3}\hat{V}_{an_i}$. SiC diodes

125



Figure B.5: FPGA control board

from CREE (C3D10065A) rated for 650V and 10A are used in the primary side diode bridges to provide benefits of zero reverse recovery current. The switches S_1 and S_2 are IXGH30N120B3 that are rated for 1200V, 30A.

2. AC-DC PETs: The diode bridges in the primary circuit are composed of (IXYS DSEE30-12A) devices. CREE (C3D10065A) diodes are used for the clamp diodes $(d_{11} \text{ and } d_{22})$. The switches S_1 and S_2 are IXGH30N120B3 that are rated for 1200V, 30A.

B.4 Gate driver

Concept 6-pack SCALE driver 6SD106E was used in the design for the matrix converter and, 2SD 106A is used for the two switches on the primary side. All of the drivers are run independently from the FPGA control board.

B.5 Sensing and control

- 1. FPGA control Board: A Xilinx XC3S200E board that was developed at the University of Minnesota is used for the control of the IGBTs (Fig. B.5). The board has on-board 12-bit ADCs for sensing voltages.
- 2. Sensor Boards: Voltage sensor board is designed that uses an LEM sensor: LV 25-P. The output of the sensor board is conditioned using an analog sallen-key

filter. This board can also be used as a current sensor using LEM LA55-P current sensor.
Appendix C

Transformer Design

The three-phase three winding transformer shown in Fig. C.1 is designed in this section for the three-phase AC-AC PET in Fig. 2.1 driving a 1kW load at 0.8 power factor. The design is done for the operating conditions in Table. C.1. the three-winding transformers are designed according to the area product method. Faraday's law given by (C.1) results

$V_{in} (\mathrm{rms})$	120 V
$V_{out} \text{ (rms)}$,	42.42 V
P_{total} (3 phases)	$1000 \mathrm{W}$
$P_{1\phi}$	333.33 W
f_s	$5~\mathrm{kHz}$
Turns ratio	1:1:1
pf	0.8

Table C.1: AC-AC PET operating conditions



Figure C.1: Three-winding transformer

in two design constraints given by (C.2) and (C.3). The values of B_{max} depends on the material and the value of core cross section area, A_c depends on the core used. Since a ferrite core is selected, $B_{max} = 5000$ gauss (0.5T). The turns ratio is selected to be 1:1:1, hence $N_1 = N_2$. The number of windings and the size of the conductors are constrained by the window area and the thermal considerations of the windings (C.4).

$$E = N \frac{d}{dt}\phi \tag{C.1}$$

$$\hat{V}_p = K_f N_1 B_{max} A_c f_s \tag{C.2}$$

$$\hat{V}_s = K_f N_2 B_{max} A_c f_s \tag{C.3}$$

$$A_w K_w J = N_1 I_{pri_1(rms)} + N_1 I_{pri_2(rms)} + N_2 I_{sec(rms)}$$
(C.4)

For this three winding transformer, $I_{pri_1(rms)}^2 + I_{pri_2(rms)}^2 = nI_{sec(rms)}^2$. Depending on the matrix converter switching, the primary windings may or may not carry the same RMS currents. In this design it is assumed that the primary winding currents are equal, $I_{pri_1(rms)} = I_{pri_2(rms)} = n \frac{I_{sec(rms)}}{\sqrt{2}}$ where, $n = N_2/N_1$. The area product is given by (C.5). Due to modulation of the matrix converter, the transformer secondary RMS current is given by (C.6), where, \hat{I}_o is the peak of the output current. The transformer design parameters are given in Table C.2. The designed transformer is in Table C.3. Thanks to Hirel Systems for winding the three transformers in Fig. C.2.

$$A_w A_c = \frac{n \hat{V_{in}}}{K_f B_{max} f K_w J} [\sqrt{2} + 1] I_{sec(rms)}$$
(C.5)

$$I_{sec(rms)} = \frac{0.866}{\sqrt{2}}\hat{I}_o \tag{C.6}$$

The transformers are characterized using an LCR meter, the leakage inductance for these transformers is approximately 15μ H and the magnetizing inductance is around 50mH.

$P_{1\phi}$	333.33 W
$\hat{I_o}$	13.99A
$I_{sec(rms)}$	8.505 A
f_s	$5 \mathrm{kH}$
Turns ratio	1:1:1
В	$0.000015 V - s/cm^2$
K_w winding factor	0.3
K_f square wave excitation	4
J	$300 \ A/cm^{2}$

Table C.2: Transformer design parameters

Table C.3: Designed transformer

Core	MAG INC 0P49928EC
A_c	$6.92 \ cm^2$
A_w	$21.223 \ cm^2$
A_p	$146.86316 \ cm^4$
$N_1 = N_2$	82 (2*AWG 17)



Figure C.2: Picture of the three three-winding transformers

Appendix D

FPGA Code for AC-AC Power Electronic Transformer

The MC based PET is controlled using a Xilinx XC3S500 FPGA. The FPGA algorithm in Fig. D.1 is written in Verilog HDL using Xilinx ISE 12.3. The pulses for S_1 and S_2 are at 10kHz and nearly 50% duty with a dead-time of 1.5µs between the transition of S_1 and S_2 . The input voltages between phase a and b and phase a and c are sensed using LV25 hall effect sensors. These sensed voltages are discretized using a 12-bit ADC and are given by V_{ab} and V_{ac} respectively. Vxi and Vyi in (D.1) and (D.2) are intermediate values that depend on the type of vector used for modulation. The input voltages are transformed into a stationary reference frame (abc $\rightarrow \alpha\beta$). The resulting stationary frame voltages are $V_{\alpha i}$ and $V_{\beta i}$. The $\alpha\beta$ voltage references for the output voltages are $V_{\alpha o}$ and $V_{\beta o}$ respectively and are generated inside the FPGA using CORDIC 12-bit sine and cosine generator.

$$Vxi = V_{ab} + V_{ac} \tag{D.1}$$

$$Vyi = \begin{cases} -V_{ab} + V_{ac} & \text{for CCW vectors} \\ V_{ab} - V_{ac} & \text{for CW vectors} \end{cases}$$
(D.2)

$$V_{\alpha i} = V x i/2 \qquad V_{\beta i} = (\sqrt{3}/2) \times V y i \tag{D.3}$$

$$|V_i| \angle \theta_i = V_{\alpha i} + j V_{\beta i} \quad |V_o| \angle \theta_o = V_{\alpha o} + j V_{\beta o} \tag{D.4}$$

When CCW rotating vectors are used, the input voltage vector in the stationary



Figure D.1: FPGA algorithm: AC-AC PET

reference frame rotates in the the same direction as the output voltage vector as shown in Fig. D.2(a). When CW vectors are used, the input and output voltage vectors rotate in opposite directions as shown in Fig. D.2(b). The *d*-axis is aligned with $V_{\alpha i}$ as shown in Fig. D.2(a) and the output voltage is transformed into this reference frame. The d-axis component of the output reference voltage is the projection of $V_{\alpha\beta o}$ on $V_{\alpha\beta i}$ i.e. $|V_{\alpha\beta o}|\cos(\theta_o - \theta_i)$ and the q-axis component of the output reference voltage is the the quadrature component i.e. $|V_{\alpha\beta o}|\sin(\theta_o - \theta_i)$. In order to optimize the number of multiplications and divisions, the calculations are done using $\cos(\theta)$ and $\sin(\theta)$ scaled by K. There is only one division operation (D.9) every switching cycle to calculate the inverse of V_i^2 .

$$\theta = \theta_o - \theta_i, \qquad K = |V_o||V_i| \tag{D.5}$$

$$V_d = K\cos(\theta) = V_{\alpha o} \times V_{\alpha i} + V_{\beta o} \times V_{\beta i}$$
(D.6)

$$V_q = K\sin(\theta) = V_{\beta o} \times V_{\alpha i} - V_{\alpha o} \times V_{\beta i}$$
(D.7)



Figure D.2: Stationary reference frame

$$V_i^2 = (1/2^2)Vxi \times Vxi + (3/2^2)Vyi \times Vyi$$
 (D.8)

$$V_{inv} = \hat{V}_{tri} / V_i^2 \tag{D.9}$$

$$V_{d/2} = V_d/2$$
 $V_{d\sqrt{3}/2} = \sqrt{3}V_d/2$ (D.10)

$$V_{q/2} = V_q/2$$
 $V_{q\sqrt{3}/2} = \sqrt{3}V_q/2$ (D.11)

D.1 Carrier Based Modulation

The modulation indices for the 6 synchronously rotating vectors when S_1 is on are given by (D.12)–(D.14); where, $\theta_i = \theta_i + \pi$ when S_2 is on. Instead of determining the angle and then performing sine and cosine operations, trigonometric identities are used to reduce the number of computations significantly, leading to saving of space in the FPGA. (D.17)–(D.18) are used in the computation of the modulation indices for CCW vectors. When S_2 is on, the magnitudes of Vxi and Vyi are the same as when S_1 is on but the sign is opposite. Hence in (D.17)–(D.18), k = 0 when S_1 is on and k = 1 when S_2 is on. V_{inv} is unaffected by the sign of Vxi and Vyi; hence, it needs to be computed only once every switching cycle. The same equations are valid for modulation indices

133

for CW vectors.

$$m_{ABC}, m_{ACB} = 1/3 + (2m/3)\cos(\theta)$$
 (D.12)

$$m_{CAB}, m_{BAC} = 1/3 + (2m/3)\cos(\theta - 2\pi/3)$$
 (D.13)

$$m_{BCA}, m_{CBA} = 1/3 + (2m/3)\cos(\theta - 4\pi/3)$$
 (D.14)

$$\theta_{ccw} = \theta_o - \theta_i \quad \theta_{cw} = \theta_o + \theta_i \tag{D.15}$$

$$m = |V_o|/|V_i| \tag{D.16}$$

$$m_{abcf}, m_{acbf} = V_{tri} + (-1^k) 2V_{inv} V_d$$
 (D.17)

$$m_{bcaf}, m_{cbaf} = V_{tri} + (-1^k) 2V_{inv}(-V_{d/2} - V_{q\sqrt{3}/2})$$
 (D.18)

$$m_{cabf}, m_{bacf} = V_{tri} + (-1^k) 2V_{inv}(-V_{d/2} + V_{q\sqrt{3}/2})$$
 (D.19)

Since, the modulation indices are scaled by $3V_{tri}$, the sawtooth carrier wave will have a peak value of $3V_{tri}$. Every $T_s/2$, the sequence of vectors applied is (ABC), (BCA) and (CAB) when CCW vectors are employed and (ACB), (CBA) and (BAC) when CW vectors are employed. These switching pulses are routed to the appropriate switches and either four-step commutation or dead-time commutation is performed. Carrier-based FPGA implementation can also be used in a MC by setting k = 0. The total of fifteen 18-bit multipliers were used in this implementation.

D.2 Space Vector Modulation

The duty-ratios for SVM are calculated only after the sector is correctly determined. Using the dq-components, the sector of operation is determined according to Fig. D.3. Once the sector is known, using trigonometric identities, the V_{d1} , V_{q1} values are transformed into the first sector according to Table D.1. The values of $V_{d/2}$, $V_{q/2}$, $V_{d\sqrt{3}/2}$ and $V_{q\sqrt{3}/2}$ that are given by (D.10)–(D.11) are calculated only once every switching cycle. V_{tri} in (D.9) is the peak of the triangular wave that the duty ratios d_{1f} and d_{2f} (D.21)– (D.20) are compared with to generate the PWM pulses for the appropriate active vector. For 120 switching cycles, vector (AAA) is selected as the zero vector for the next 120 cycles (BBB) is selected and the following 120 cycles (CCC) is used. Depending on the sector and kind of vector (clockwise or counter-clockwise) to be applied, these switching



Figure D.3: SVM: Sector Determination

pulses are routed appropriately. All multiplications and divisions by 2 are performed by shift operations. The FPGA implementation uses sixteen 18-bit multipliers in the FPGA.

Sector	V_{d1}	V_{q1}
1	V_d	V_q
2	$V_{d/2} + V_{q\sqrt{3}/2}$	$V_{q/2} - V_{d\sqrt{3}/2}$
3	$-V_{d/2} + V_{q\sqrt{3}/2}$	$-V_{q/2}-V_{d\sqrt{3}/2}$
4	$-V_d$	$-V_q$
5	$-V_{d/2} - V_{q\sqrt{3}/2}$	$-V_{q/2} + V_{d\sqrt{3}/2}$
6	$V_{d/2} - V_{q\sqrt{3}/2}$	$V_{q/2} + V_{d\sqrt{3}/2}$

Table D.1: Sector Transformation

$$d_{2f} = V_{inv} \times V_{a1} \times (\sin \pi/3)^{-1}$$
(D.20)

$$d_{1f} = V_{inv} \times V_{d1} - d_{2f}/2 \tag{D.21}$$

D.3 Commutation

The commutation of switches is done either using dead-time commutation or four-step commutation. The currents for phase u and phase v are sensed at 100kHz using sensors LA55-P. The current for phase w is determined from phase u and v currents. If the direction of phase currents are accurately known, four-step commutation is performed



Figure D.4: Four-step commutation pulses: C1: S_{Au1} ; C2: S_{Au2} ; C4: ideal pulse

else dead-time commutation is performed. In this code, the dead-time (t_{dt}) is set to 1.5μ s. The time for each step in four-step commutation is 500ns. The gate pulses for four-step commutation of S_{Au1} and S_{Au2} are shown in Fig. D.4. Thanks to Rohit Baranwal for the verilog state-machine for four-step commutation.

Note: Parts of this chapter have been reprinted from [60]

Appendix E

FPGA Code for AC-DC Power Electronic Transformer

The single-phase and three-phase AC-DC PETs are controlled using a Xilinx XC3S500E FPGA. In entire switching algorithm is written in Verilog HDL.

E.1 Single-phase AC-DC Power Electronic Transformer

The input and output voltages, V_i and V_o are sensed using LV-25 voltage sensors and their values discretized by 12-bit ADCs. The ADCs sample these voltages at 10kHz.

The pulses for S_1 and S_2 are at 10kHz and nearly 50% duty with a dead-time of 1.5μ s between the transition of S_1 and S_2 . Xilinx CORE Generator division block is used to perform a 16-bit division on V_o . The inverse of V_o is multiplied with the input to generate a duty ratio given by (E.2). A saw-tooth carrier wave counts from 0 to V_{saw} every 50μ s as shown in Fig. E.1. This sawtooth waveform is compared with the duty ratio to generate phase shifted PWM pulses. Depending on the polarity of the input voltage, the pulses are routed to the appropriate switches. A dead-time of 1.5μ s is introduced between the switching pulses of the switches in the same leg of the secondary side converter. The modulation is suspended in this converter for the following cases : 1) If $V_{dc} < V_i$ 2) If the phase shift is too high $\frac{V_{saw}}{2} - \frac{d}{2} + \delta \leq 0$ or if $\frac{V_{saw}}{2} + \frac{d}{2} + \delta > V_{saw}$.

3) If d is too small.

$$V_{inv} = \frac{V_{saw}}{V_o} \tag{E.1}$$

$$d = V_{inv} \times V_i \tag{E.2}$$

(E.3)



Figure E.1: FPGA implementation: Single-phase AC-DC PET

E.2 Three-phase AC-DC Power Electronic Transformer

A brief description of the algorithm shown in Fig. E.2 is given in this section.



Figure E.2: FPGA algorithm: Three-phase AC-DC PET

The input voltages, V_{ab} , V_{ac} and V_{dc} are sampled at 10kHz using 12-bit ADCs. The division, V_{saw}/V_{dc} is done using Xilinx CORE Generator 16 bit division.

The input AC voltages V_{ab} and V_{ac} are converted to dq-stationary reference frame using (E.4)- (E.6). The values of $V_{d/2}$, $V_{q/2}$, $V_{d\sqrt{3}/2}$ and $V_{q\sqrt{3}/2}$ that are given by (D.10)– (D.11) are calculated only once every switching cycle. The sector can be determined by checking the sign of *wave* (E.7), V_d and V_q as shown in Fig. D.3. Just as in the AC-AC SVM implementation, the values of of V_d and V_q are transformed into Sector 1 and are given by Table. D.1.

$$Vxi = V_{ab} + V_{ac} \tag{E.4}$$

$$Vyi = -V_{ab} + V_{ac} \tag{E.5}$$

$$V_d = \frac{Vxi}{2} \qquad V_q = \frac{\sqrt{3}}{2} \times Vyi \tag{E.6}$$

$$wave = \sqrt{3}|V_d| - |V_q|$$

= $\frac{\sqrt{3}}{2}(|Vxi| - |Vyi|)$ (E.7)

In each sector, once V_{d1} and V_{q1} are known, the duty ratios for the adjacent vectors can be calculated using (D.20) and (D.21). The duty ratios are routed to the PWM block where the the phase-shifted PWM pulses for the switches are generated as shown in Fig. E.1. A dead-time of 1.5μ s is introduced between the switching transitions in an inverter leg. Depending on the sector and switching state of the primary side switches S_1 and S_2 , these PWM pulses are routed to the correct switches.

The selected FPGA has twenty dedicated 18-bit hardware multipliers. In this code only six hardware multipliers are used.